% EJC papers *must* begin with the following two lines. 
\documentclass[12pt]{article}
\usepackage{e-jc}
\specs{P1.40}{25(1)}{2018}

% Please remove all other commands that change parameters such as
% margins or pagesizes.

% only use standard LaTeX packages
% only include packages that you actually need

% we recommend these ams packages
\usepackage{amsthm,amsmath,amssymb,latexsym}

% we recommend the graphicx package for importing figures
\usepackage{graphicx}

% use this command to create hyperlinks (optional and recommended)
%\usepackage[colorlinks=true,citecolor=black,linkcolor=black,urlcolor=blue]{hyperref}

% use these commands for typesetting doi and arXiv references in the bibliography
\newcommand{\doi}[1]{\href{http://dx.doi.org/#1}{\texttt{doi:#1}}}
\newcommand{\arxiv}[1]{\href{http://arxiv.org/abs/#1}{\texttt{arXiv:#1}}}

% all overfull boxes must be fixed; 
% i.e. there must be no text protruding into the margins


% declare theorem-like environments
\theoremstyle{plain}
\newtheorem{thm}{Theorem}[section]
\newtheorem{refthm}[thm]{Theorem}
\newtheorem{Thm}{Theorem}
\renewcommand{\theThm}{\Alph{Thm}}
\newtheorem{cor}[thm]{Corollary}
\newtheorem{prop}[thm]{Proposition}
\newtheorem{lem}[thm]{Lemma}
\newtheorem{obs}[thm]{Observation}
\newtheorem{Q}[thm]{Question} 
\renewcommand{\theQ}{}
\newtheorem{claim}{Claim}[section] 
\newtheorem{subclaim}{Subclaim}[claim]
\newtheorem{fact}[claim]{Fact} 
\newtheorem{con}{Conjecture}
\newtheorem{pro}{Problem}
\numberwithin{equation}{section}
\newtheorem{remark}{\normalfont\itshape Remark}

\def\A{{ \mathcal{A}}}
\def\B{{ \mathcal{B}}}
\def\C{{ \mathcal{C}}}
\def\D{{ \mathcal{D}}}
\def\F{{ \mathcal{F}}}
\def\G{{ \mathcal{G}}}
\def\H{{ \mathcal{H}}}
\def\K{{ \mathcal{K}}}
\def\L{{ \mathcal{L}}}
\def\P{{ \mathcal{P}}}
\def\Q{{ \mathcal{Q}}}
\def\R{{ \mathcal{R}}}
\def\S{{ \mathcal{S}}}
\def\T{{ \mathcal{T}}}


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

% if needed include a line break (\\) at an appropriate place in the title

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\title{\bf The existence of a path-factor\\without small odd paths}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\author{Yoshimi Egawa\\
\small Department of Applied Mathematics\\[-0.8ex]
\small Tokyo University of Science\\[-0.8ex]
\small 1-3 Kagurazaka, Shinjuku-ku, Tokyo 162-8601, Japan\\
\small \tt disc\_student\_seminar@rs.tus.ac.jp\\
\and
Michitaka Furuya\thanks{Partly supported by Japan Society for the Promotion of Science, Grant-in-Aid for Young Scientists (B) 26800086}\\
\small College of Liberal Arts and Science\\[-0.8ex]
\small Kitasato University\\[-0.8ex]
\small 1-15-1 Kitasato, Minami-ku, Sagamihara, Kanagawa 252-0373, Japan\\
\small\tt michitaka.furuya@gmail.com\\
}

% \date{\dateline{submission date}{acceptance date}\\
% \small Mathematics Subject Classifications: comma separated list of
% MSC codes available from http://www.ams.org/mathscinet/freeTools.html}

\date{\dateline{Dec 12, 2015}{Dec 5, 2017}{Mar 2, 2018}\\
\small Mathematics Subject Classifications: 05C70}

\begin{document}
\begin{thebibliography}{10}

\bibitem{AAE}
J.~Akiyama, D.~Avis and H.~Era,
On a $\{1,2\}$-factor of a graph,
TRU Math. \textbf{16} (1980) 97--102.

\bibitem{D}
R.~Diestel,
``Graph Theory'' (4th edition), Graduate Texts in Mathematics \textbf{173},
Springer (2010).

\bibitem{EF}
Y.~Egawa and M.~Furuya,
Path-factors involving paths of order seven and nine,
Theory Appl. Graphs \textbf{3(1)} (2016) \#5.

\bibitem{EFO}
Y.~Egawa, M.~Furuya and K.~Ozeki,
Sufficient conditions for the existence of a path-factor which are related to odd components,
preprint.

\bibitem{K}
A.~Kaneko,
A necessary and sufficient condition for the existence of a path factor every component of which is a path of length at least two,
J. Combin. Theory Ser. B \textbf{88} (2003) 195--218.

\bibitem{KLS}
M.~Kano, C.~Lee and K.~Suzuki,
Path and cycle factors of cubic bipartite graphs,
Discuss. Math. Graph Theory \textbf{28} (2008) 551--556.

\bibitem{KMOO}
K.~Kawarabayashi, H.~Matsuda, Y.~Oda and K.~Ota,
Path factors in cubic graphs,
J. Graph Theory \textbf{39} (2002) 188--193.

\bibitem{LP}
M.~Loebl and S.~Poljak,
Efficient subgraph packing,
J. Combin. Theory Ser. B \textbf{59} (1993) 106--121.

\end{thebibliography}


\maketitle


\begin{abstract}
A $\{P_{2},P_{5}\}$-factor of a graph is a spanning subgraph of the graph each of whose components is isomorphic to either $P_{2}$ or $P_{5}$, where $P_{n}$ denote the path of order $n$.
In this paper, we show that if a graph $G$ satisfies $c_{1}(G-X)+\frac{2}{3}c_{3}(G-X)\leq \frac{4}{3}|X|+\frac{1}{3}$ for all $X\subseteq V(G)$, then $G$ has a $\{P_{2},P_{5}\}$-factor, where $c_{i}(G-X)$ is the number of components $C$ of $G-X$ with $|V(C)|=i$.
Moreover, it is shown that above condition is sharp.

  % keywords are optional
  \bigskip\noindent \textbf{Keywords:} path-factor; component-factor; matching.
\end{abstract}



%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Introduction}\label{sec1}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

In this paper, all graphs are finite and simple.
Let $G$ be a graph.
Let $V(G)$ and $E(G)$ denote the vertex set and the edge set of $G$, respectively.
The {\it order} of $G$ is the cardinality $|V(G)|$ of $V(G)$.
%For $u\in V(G)$, we let $N_{G}(u)$ and $d_{G}(u)$ denote the {\it neighborhood} and the {\it degree} of $u$, respectively.
For $u\in V(G)$, the {\it neighborhood} of $u$, denoted by $N_{G}(u)$, is the set of vertices adjacent to $u$, and the {\it degree} of $u$, denoted by $d_{G}(u)$, is the number of vertices adjacent to $u$; thus $N_{G}(u)=\{v\in V(G)\mid uv\in E(G)\}$ and $d_{G}(u)=|N_{G}(u)|$.
For $U\subseteq V(G)$, let $N_{G}(U)=(\bigcup _{u\in U}N_{G}(u))-U$.
For disjoint sets $X,Y\subseteq V(G)$, let $E_{G}(X,Y)$ denote the set of edges of $G$ joining a vertex in $X$ and a vertex in $Y$.
For $X\subseteq V(G)$, let $G[X]$ denote the subgraph of $G$ induced by $X$.
%For a graph $H$ and an integer $s\geq 2$, we let $sH$ denote the disjoint union of $s$ copies of $H$.
%For two graphs $H_{1}$ and $H_{2}$, we let $H_{1}+H_{2}$ denote the {\it join} of $H_{1}$ and $H_{2}$.
If $G$ is isomorphic to a graph $H$, we write $G\simeq H$.
For two vertex-disjoint graphs $H_{1}$ and $H_{2}$, the {\it join} of $H_{1}$ and $H_{2}$, denoted by $H_{1}+H_{2}$, is the graph obtained from $H_{1}$ and $H_{2}$ by joining each vertex of $H_{1}$ to all vertices of $H_{2}$.
Let $P_{n}$ denote the {\it path} of order $n$.
For terms and symbols not defined here, we refer the reader to \cite{D}.

A subgraph of a graph $G$ is {\it spanning} if the subgraph contains all vertices of $G$.
For a set $\H$ of connected graphs, a spanning subgraph $F$ of a graph is called an {\it $\H$-factor} if each component of $F$ is isomorphic to a graph in $\H$.
A {\it path-factor} of a graph is a spanning subgraph whose components are paths of order at least $2$.
Since every path of order at least $4$ can be partitioned into paths of orders $2$ and $3$, a graph has a path-factor if and only if it has a $\{P_{2},P_{3}\}$-factor.
Akiyama, Avis and Era~\cite{AAE} gave a necessary and sufficient condition for the existence of a path-factor (here $i(G)$ denotes the number of isolated vertices of a graph $G$).

\begin{Thm}[Akiyama, Avis and Era~\cite{AAE}]%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\label{ThmA}
A graph $G$ has a $\{P_{2},P_{3}\}$-factor if and only if $i(G-X)\leq 2|X|$ for all $X\subseteq V(G)$.
\end{Thm}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

Now we consider a path-factor with additional conditions.
For example, one may require a path-factor to consist of components of large order.
Concerning such a problem, Kaneko~\cite{K} gave a necessary and sufficient condition for the existence of a path-factor whose components have order at least $3$.
On the other hand, for $k\geq 4$, it is not known that whether the existence problem of a path-factor whose components have order at least $k$ is polynomially solvable or not, though some results about such a factor have been obtained (see, for example, Kano, Lee and Suzuki~\cite{KLS} and Kawarabayashi, Matsuda, Oda and Ota~\cite{KMOO}).

In this paper, we study a different type of path-factor problem.
Specifically, we focus on the existence of a $\{P_{2},P_{2k+1}\}$-factor ($k\geq 2$).

%There are two motivations to study such factors.
The motivation to study such factors is related the notion of a hypomatchable graph.
A graph $H$ is {\it hypomatchable} if $H-x$ has a perfect matching for every $x\in V(H)$.
A graph is a {\it propeller} if it is obtained from a hypomatchable graph $H$ by adding new vertices $a,b$ together with edge $ab$, and joining $a$ to some vertices of $H$.
Loebal and Poljak~\cite{LP} proved the following theorem.

\begin{Thm}[Loebal and Poljak~\cite{LP}]%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\label{ThmB}
Let $H$ be a connected graph.
If either $H$ has a perfect matching, or $H$ is hypomatchable, or $H$ is a propeller, then the existence problem of a $\{P_{2},H\}$-factor is polynomially solvable.
The problem is {\bf NP}-complete for all other graphs $H$.
\end{Thm}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

In particular, for $k\geq 2$, the existence problem of a $\{P_{2},P_{2k+1}\}$-factor is {\bf NP}-complete.
Because of this fact, existence problems concerning $\{P_{2},P_{2k+1}\}$-factors seem to have unjustly been ignored.
However, in general, the fact that a problem is {\bf NP}-complete in terms of algorithm may mean that one cannot obtain a necessary and sufficient condition, but it might be possible to obtain sufficient conditions.
%in terms of algorithm does not mean that one cannot obtain a theoretical result concerning the problem.
From this viewpoint, in this paper, we prove a theorem on the existence of a $\{P_{2},P_{5}\}$-factor which, we hope, will serve as an initial attempt to develop the theory of $\{P_{2},P_{2k+1}\}$-factors.

%The other motivation is the fact that a $\{P_{2},P_{2k+1}\}$-factor is a useful tool for finding large matchings.
%It is easy to see that if a graph $G$ has a $\{P_{2},P_{2k+1}\}$-factor, then $G$ has a matching $M$ with $|M|\geq \frac{k}{2k+1}|V(G)|$.
%Thus the existence of a $\{P_{2},P_{2k+1}\}$-factor helps to find large matchings.
%, and hence we cannot expect a manageable necessary and sufficient condition for it.
%Instead of that, we seek a better sufficient condition for it.

In order to state our theorem, we need some more definitions.
For a graph $H$, let $\C(H)$ be the set of components of $H$, and for $i\geq 1$, let $\C _{i}(H)=\{C\in \C(H)\mid |V(C)|=i\}$ and $c_{i}(H)=|\C _{i}(H)|$.
Note that $c_{1}(H)$ is the number of isolated vertices of $H$ (i.e., $c_{1}(H)=i(H)$).
If a graph $G$ has a $\{P_{2},P_{5}\}$-factor, then $c_{1}(G-X)+\frac{1}{2}c_{3}(G-X)\leq \frac{3}{2}|X|$ for all $X\subseteq V(G)$ (see Section~\ref{sec-nec}).
Thus if a condition concerning $c_{1}(G-X)$ and $c_{3}(G-X)$ for $X\subseteq V(G)$ assures us the existence of a $\{P_{2},P_{5}\}$-factor, then it will make a useful sufficient condition.

The main purpose of this paper is to prove the following theorem.

\begin{thm}%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\label{thm1}
Let $G$ be a graph.
If $c_{1}(G-X)+\frac{2}{3}c_{3}(G-X)\leq \frac{4}{3}|X|+\frac{1}{3}$ for all $X\subseteq V(G)$, then $G$ has a $\{P_{2},P_{5}\}$-factor.
\end{thm}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

We prove Theorem~\ref{thm1} in Sections~\ref{sec2} and \ref{sec3}.
In Subsection~\ref{sec4.1}, we show that the bound $\frac{4}{3}|X|+\frac{1}{3}$ in Theorem~\ref{thm1} is best possible.

In our proof of Theorem~\ref{thm1}, we make use of the following fact.

\begin{fact}%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\label{fact1}
A graph $G$ has a $\{P_{2},P_{5}\}$-factor if and only if $G$ has a path-factor $F$ with $\C_{3}(F)=\emptyset $.
\end{fact}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

We conclude this section with a conjecture concerning $\{P_{2},P_{2k+1}\}$-factors with $k\geq 3$.
If $k\geq 3$ and $k\equiv 0~(\mbox{mod }3)$, then there exist infinitely many graphs $G$ having no $\{P_{2},P_{2k+1}\}$-factor such that $\sum _{0\leq i\leq k-1}c_{2i+1}(G-X)\leq \frac{4k+6}{8k+3}|X|+\frac{2k+3}{8k+3}$ for all $X\subseteq V(G)$ (see Subsection~\ref{sec4.2}).
Thus we pose the following conjecture.\footnote{Recently the conjecture for the case where $k\in \{3,4\}$ was affirmatively settled in \cite{EF}.
On the other hand, counterexamples to the conjecture for the case where $k\geq 36$ were constructed in \cite{EFO} (see Section~\ref{sec5}).}


\begin{con}%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\label{con1}
Let $k\geq 3$, and let $G$ be a graph.
If $\sum _{0\leq i\leq k-1}c_{2i+1}(G-X)\leq \frac{4k+6}{8k+3}|X|$ for all $X\subseteq V(G)$, then $G$ has a $\{P_{2},P_{2k+1}\}$-factor.
\end{con}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%




%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{A necessary condition for a $\{P_{2},P_{5}\}$-factor}\label{sec-nec}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

In this section, we give a necessary condition for the existence of a $\{P_{2},P_{5}\}$-factor in terms of invariants $c_{1}$ and $c_{3}$.
We show the following proposition.

\begin{prop}%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\label{prop1.1}
If a graph $G$ has a $\{P_{2},P_{5}\}$-factor, then $c_{1}(G-X)+\frac{1}{2}c_{3}(G-X)\leq \frac{3}{2}|X|$ for all $X\subseteq V(G)$.
\end{prop}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\proof
Let $F$ be a $\{P_{2},P_{5}\}$-factor of $G$, and let $X\subseteq V(G)$.
It can be verified that
\begin{align}
c_{1}(P-X)+\frac{1}{2}c_{3}(P-X)\leq \frac{3}{2}|V(P)\cap X| \mbox{ for every } P\in \C(F).\label{sec1-1}
\end{align}
Since every component $C$ of $G-X$ with $|V(C)|=1$ belongs to $\C_{1}(F-X)$, we have
\begin{align}
|&\C_{1}(G-X)|=|\C_{1}(F-X)|-|\C_{1}(F-X)-\C_{1}(G-X)|.\label{sec1-2}
\end{align}
Furthermore,
\begin{align}
|&\C_{3}(G-X)|\leq |\C_{3}(F-X)|+|\C_{3}(G-X)-\C_{3}(F-X)|.\label{sec1-3}
\end{align}

Let $C$ be a component of $G-X$ with $|V(C)|=3$ which does not belong to $\C_{3}(F-X)$.
Then $C$ intersects with at least two components of $F-X$.
Since $|V(C)|=3$, $C$ contains a component of $F-X$ of order $1$.
Since $C$ is arbitrary, this implies that
\begin{align}
|\C_{3}(G-X)-\C_{3}(F-X)|\leq |\C_{1}(F-X)-\C_{1}(G-X)|.\label{sec1-4}
\end{align}
By (\ref{sec1-1})--(\ref{sec1-4}),
\begin{align}
c_{1}(G-X)&+\frac{1}{2}c_{3}(G-X)\nonumber \\
&\leq (|\C_{1}(F-X)|-|\C_{1}(F-X)-\C_{1}(G-X)|)\nonumber \\
&\quad +\frac{1}{2}(|\C_{3}(F-X)|+|\C_{3}(G-X)-\C_{3}(F-X)|)\nonumber \\
&\leq (|\C_{1}(F-X)|-|\C_{1}(F-X)-\C_{1}(G-X)|)\nonumber \\
&\quad +\frac{1}{2}(|\C_{3}(F-X)|+|\C_{1}(F-X)-\C_{1}(G-X)|)\nonumber \\
&\leq |\C_{1}(F-X)|+\frac{1}{2}|\C_{3}(F-X)|\nonumber \\
&= \sum _{P\in \C(F)}\left(c_{1}(P-X)+\frac{1}{2}c_{3}(P-X)\right)\nonumber \\
&\leq \frac{3}{2}\sum _{P\in \C(F)}|V(P)\cap X|\nonumber \\
&= \frac{3}{2}|X|.\nonumber
\end{align}
Thus we get the desired conclusion.
\qed





%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{A path-factor in bipartite graph}\label{sec2}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

Let $G$ be a bipartite graph with bipartition $(S,T)$.
A subgraph $F$ of $G$ is {\it $S$-central} if $S\subseteq V(F)$ and $|V(A)\cap T|\geq |V(A)\cap S|$ for every $A\in \C(F)$.

In this section, we focus on the existence of a special path-factor in bipartite graphs, and show the following theorem, which will be used in our proof of Theorem~\ref{thm1}.

\begin{thm}%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\label{thm2}
Let $S$, $T_{1}$ and $T_{2}$ be disjoint sets with $1\leq |S|\leq |T_{1}|+|T_{2}|$ and $|T_{1}|+\frac{2}{3}|T_{2}|\leq \frac{4}{3}|S|+\frac{1}{3}$, and set $T=T_{1}\cup T_{2}$.
Let $G$ be a bipartite graph with bipartition $(S,T)$ satisfying the property that for every $X\subseteq S$, we have either $|N_{G}(X)\cap T_{1}|+\frac{2}{3}|N_{G}(X)\cap T_{2}|\geq \frac{4}{3}|X|$ or $N_{G}(X)=T$.
Then $G$ has an $S$-central path-factor $F$ such that $V(A)\cap T_{2}\not=\emptyset $ for every $A\in \C_{3}(F)$.
\end{thm}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

Our proof of Theorem~\ref{thm2} is rather technical, and thus readers not interested in technical details are advised to skip the rest of Section~\ref{sec2} and proceed to Section~\ref{sec3}.
Before proving the theorem, we prove a lemma.

\begin{lem}%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\label{lem2.1}
Let $S$, $T_{1}$, $T_{2}$, $T$ and $G$ be as in Theorem~\ref{thm2}.
Then $G$ has an $S$-central path-factor.
\end{lem}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\proof
Let $X\subseteq S$.
If $|N_{G}(X)\cap T_{1}|+\frac{2}{3}|N_{G}(X)\cap T_{2}|\geq \frac{4}{3}|X|$, then $|N_{G}(X)|\geq |N_{G}(X)\cap T_{1}|+\frac{2}{3}|N_{G}(X)\cap T_{2}|\geq \frac{4}{3}|X|\geq |X|$; if $N_{G}(X)=T_{1}\cup T_{2}$, then $|N_{G}(X)|=|T_{1}|+|T_{2}|\geq |S|\geq |X|$.
In either case, we have $|N_{G}(X)|\geq |X|$.
Since $X$ is arbitrary, $G$ has a matching covering $S$ by Hall's marriage theorem.
In particular, $G$ has an $S$-central subgraph $F$ such that every component of $F$ is a path of order at least $2$.
Choose $F$ so that $|V(F)|$ is as large as possible.

Suppose that $V(G)-V(F)\not=\emptyset $.
Note that $V(G)-V(F)\subseteq T$.
Now we define the set $\A$ of components of $F$ as follows:
Let $\A_{1}$ be the set of components $A$ of $F$ with $E_{G}(V(A)\cap S,V(G)-V(F))\not=\emptyset $.
For each $i\geq 2$, let $\A_{i}$ be the set of components $A$ of $F$ with $A\not\in \bigcup _{1\leq j\leq i-1}\A_{j}$ and $E_{G}(V(A)\cap S,\bigcup _{A'\in \A_{i-1}}(V(A')\cap T))\not=\emptyset $.
Let $\A=\bigcup _{i\geq 1}\A_{i}$.



\begin{claim}%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\label{cl2.1.1}
Every path belonging to $\A$ is isomorphic to $P_{3}$.
\end{claim}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\proof
Suppose that $\A$ contains a path which is not isomorphic to $P_{3}$.
Let $i$ be the minimum integer such that $\A_{i}$ contains a path $A_{i}=v^{(i)}_{1}\cdots v^{(i)}_{l}$ with $A_{i}\not\simeq P_{3}$.
By the minimality of $i$, every path belonging to $\bigcup _{1\leq j\leq i-1}\A_{j}$ is isomorphic to $P_{3}$.
Hence by the definition of $\A_{j}$, there exists a vertex $v^{(0)}_{1}\in V(G)-V(F)$ and there exist paths $A_{j}=v^{(j)}_{1}v^{(j)}_{2}v^{(j)}_{3}\in \A_{j}~(1\leq j\leq i-1)$ such that $E_{G}(V(A_{1})\cap S,\{v^{(0)}_{1}\})\not=\emptyset $ and $E_{G}(V(A_{j+1})\cap S,V(A_{j})\cap T)\not=\emptyset $ for every $j~(1\leq j\leq i-1)$.
For each $j~(1\leq j\leq i-1)$, by renumbering the veritices $v^{(j)}_{1},v^{(j)}_{2},v^{(j)}_{3}$ of $A_{j}$ backward (i.e., by tracing the path $v^{(j)}_{1}v^{(j)}_{2}v^{(j)}_{3}$ backward and numbering the vertices accordingly) if necessary, we may assume that $E_{G}(V(A_{j+1})\cap S,\{v^{(j)}_{1}\})\not=\emptyset $.
Let $m$ be an index such that $v^{(i)}_{m}v^{(i-1)}_{1}\in E(G)$.
Note that $l\geq 2$ and $l\not=3$.
Thus by renumbering the vertices $v^{(i)}_{1},\ldots,v^{(i)}_{l}$ of $A_{i}$ backward if necessary, we may assume that $m\not=2$ if $l$ is odd, and $m$ is odd if $l$ is even.
Let $B_{j}=v^{(j-1)}_{1}v^{(j)}_{2}v^{(j)}_{3}~(1\leq j\leq i-1)$, $B_{i}=v^{(i-1)}_{1}v^{(i)}_{m}v^{(i)}_{m+1}\cdots v^{(i)}_{l}$ and $B_{i+1}=v^{(i)}_{1}\cdots v^{(i)}_{m-1}$ (see Figure~\ref{claim3.1fig1}).
Note that $B_{i+1}=\emptyset $ if and only if $l$ is even and $m=1$.
Then $|V(B_{j})\cap T|\geq |V(B_{j})\cap S|$ for every $j~(1\leq j\leq i+1)$.
Therefore $F'=(F-(\bigcup _{1\leq j\leq i}V(A_{j})))\cup (\bigcup _{1\leq j\leq i+1}B_{j})$ is an $S$-central subgraph of $G$ such that $V(F')=V(F)\cup \{v^{(0)}_{1}\}$ and every component of $F'$ is a path of order at least $2$, which contradicts the maximality of $F$.
\qed


\begin{figure}[h]
\begin{center}
%WinTpicVersion4.28b
{\unitlength 0.1in
\begin{picture}( 29.0000, 25.2900)(  9.4000,-37.2700)
% CIRCLE 2 0 0 0 Black Black
% 4 995 1400 995 1450 995 1450 995 1450
% 
\special{sh 1.000}%
\special{ia 996 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 996 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1095 1800 1095 1850 1095 1850 1095 1850
% 
\special{sh 1.000}%
\special{ia 1096 1800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 1096 1800 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1195 1400 1195 1450 1195 1450 1195 1450
% 
\special{sh 1.000}%
\special{ia 1196 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 1196 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1395 1400 1395 1450 1395 1450 1395 1450
% 
\special{sh 1.000}%
\special{ia 1396 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 1396 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1595 1400 1595 1450 1595 1450 1595 1450
% 
\special{sh 1.000}%
\special{ia 1596 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 1596 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1295 1800 1295 1850 1295 1850 1295 1850
% 
\special{sh 1.000}%
\special{ia 1296 1800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 1296 1800 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1495 1800 1495 1850 1495 1850 1495 1850
% 
\special{sh 1.000}%
\special{ia 1496 1800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 1496 1800 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1795 1400 1795 1450 1795 1450 1795 1450
% 
\special{sh 1.000}%
\special{ia 1796 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 1796 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1695 1800 1695 1850 1695 1850 1695 1850
% 
\special{sh 1.000}%
\special{ia 1696 1800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 1696 1800 50 50  0.0000000  6.2831853}%
% LINE 0 0 3 0 Black Black
% 2 995 1400 1095 1800
% 
\special{pn 20}%
\special{pa 996 1400}%
\special{pa 1096 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1195 1400 1295 1800
% 
\special{pn 20}%
\special{pa 1196 1400}%
\special{pa 1296 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1395 1400 1495 1800
% 
\special{pn 20}%
\special{pa 1396 1400}%
\special{pa 1496 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1595 1400 1695 1800
% 
\special{pn 20}%
\special{pa 1596 1400}%
\special{pa 1696 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1795 1400 1695 1800
% 
\special{pn 20}%
\special{pa 1796 1400}%
\special{pa 1696 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1595 1400 1495 1800
% 
\special{pn 20}%
\special{pa 1596 1400}%
\special{pa 1496 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1395 1400 1295 1800
% 
\special{pn 20}%
\special{pa 1396 1400}%
\special{pa 1296 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1195 1400 1095 1800
% 
\special{pn 20}%
\special{pa 1196 1400}%
\special{pa 1096 1800}%
\special{fp}%
% CIRCLE 2 0 0 0 Black Black
% 4 1990 1400 1990 1450 1990 1450 1990 1450
% 
\special{sh 1.000}%
\special{ia 1990 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 1990 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 2090 1800 2090 1850 2090 1850 2090 1850
% 
\special{sh 1.000}%
\special{ia 2090 1800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 2090 1800 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 2190 1400 2190 1450 2190 1450 2190 1450
% 
\special{sh 1.000}%
\special{ia 2190 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 2190 1400 50 50  0.0000000  6.2831853}%
% LINE 0 0 3 0 Black Black
% 2 1990 1400 2090 1800
% 
\special{pn 20}%
\special{pa 1990 1400}%
\special{pa 2090 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 2190 1400 2090 1800
% 
\special{pn 20}%
\special{pa 2190 1400}%
\special{pa 2090 1800}%
\special{fp}%
% DOT 0 0 3 0 Black Black
% 4 3000 1600 2900 1600 3100 1600 3100 1600
% 
\special{pn 4}%
\special{sh 1}%
\special{ar 3000 1600 16 16 0  6.28318530717959E+0000}%
\special{sh 1}%
\special{ar 2900 1600 16 16 0  6.28318530717959E+0000}%
\special{sh 1}%
\special{ar 3100 1600 16 16 0  6.28318530717959E+0000}%
\special{sh 1}%
\special{ar 3100 1600 16 16 0  6.28318530717959E+0000}%
% STR 2 0 3 0 Black Black
% 4 1400 1900 1400 2000 5 0 0 0
% $A_{i}$
\put(14.0000,-20.0000){\makebox(0,0){$A_{i}$}}%
% STR 2 0 3 0 Black Black
% 4 2095 1900 2095 2000 5 0 0 0
% $A_{i-1}$
\put(20.9500,-20.0000){\makebox(0,0){$A_{i-1}$}}%
% CIRCLE 2 0 0 0 Black Black
% 4 2390 1400 2390 1450 2390 1450 2390 1450
% 
\special{sh 1.000}%
\special{ia 2390 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 2390 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 2490 1800 2490 1850 2490 1850 2490 1850
% 
\special{sh 1.000}%
\special{ia 2490 1800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 2490 1800 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 2590 1400 2590 1450 2590 1450 2590 1450
% 
\special{sh 1.000}%
\special{ia 2590 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 2590 1400 50 50  0.0000000  6.2831853}%
% LINE 0 0 3 0 Black Black
% 2 2390 1400 2490 1800
% 
\special{pn 20}%
\special{pa 2390 1400}%
\special{pa 2490 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 2590 1400 2490 1800
% 
\special{pn 20}%
\special{pa 2590 1400}%
\special{pa 2490 1800}%
\special{fp}%
% STR 2 0 3 0 Black Black
% 4 2495 1900 2495 2000 5 0 0 0
% $A_{i-2}$
\put(24.9500,-20.0000){\makebox(0,0){$A_{i-2}$}}%
% CIRCLE 2 0 0 0 Black Black
% 4 3390 1400 3390 1450 3390 1450 3390 1450
% 
\special{sh 1.000}%
\special{ia 3390 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 3390 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3490 1800 3490 1850 3490 1850 3490 1850
% 
\special{sh 1.000}%
\special{ia 3490 1800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 3490 1800 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3590 1400 3590 1450 3590 1450 3590 1450
% 
\special{sh 1.000}%
\special{ia 3590 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 3590 1400 50 50  0.0000000  6.2831853}%
% LINE 0 0 3 0 Black Black
% 2 3390 1400 3490 1800
% 
\special{pn 20}%
\special{pa 3390 1400}%
\special{pa 3490 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 3590 1400 3490 1800
% 
\special{pn 20}%
\special{pa 3590 1400}%
\special{pa 3490 1800}%
\special{fp}%
% STR 2 0 3 0 Black Black
% 4 3495 1900 3495 2000 5 0 0 0
% $A_{1}$
\put(34.9500,-20.0000){\makebox(0,0){$A_{1}$}}%
% CIRCLE 2 0 0 0 Black Black
% 4 3790 1400 3790 1450 3790 1450 3790 1450
% 
\special{sh 1.000}%
\special{ia 3790 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 3790 1400 50 50  0.0000000  6.2831853}%
% SPLINE 2 1 3 0 Black Black
% 6 1490 1800 1490 1600 1490 1400 1690 1200 2000 1400 2000 1400
% 
\special{pn 8}%
\special{pn 8}%
\special{pa 1490 1800}%
\special{pa 1493 1737}%
\special{fp}%
\special{pa 1494 1673}%
\special{pa 1494 1672}%
\special{pa 1492 1610}%
\special{fp}%
\special{pa 1486 1546}%
\special{pa 1482 1512}%
\special{pa 1480 1483}%
\special{fp}%
\special{pa 1486 1419}%
\special{pa 1486 1416}%
\special{pa 1496 1384}%
\special{pa 1506 1360}%
\special{fp}%
\special{pa 1540 1306}%
\special{pa 1548 1296}%
\special{pa 1570 1270}%
\special{pa 1582 1259}%
\special{fp}%
\special{pa 1634 1222}%
\special{pa 1652 1212}%
\special{pa 1680 1202}%
\special{pa 1692 1200}%
\special{fp}%
\special{pa 1756 1204}%
\special{pa 1766 1206}%
\special{pa 1792 1216}%
\special{pa 1814 1228}%
\special{fp}%
\special{pa 1865 1266}%
\special{pa 1872 1272}%
\special{pa 1900 1294}%
\special{pa 1913 1307}%
\special{fp}%
\special{pa 1957 1354}%
\special{pa 1978 1376}%
\special{pa 2000 1400}%
\special{fp}%
% SPLINE 2 1 3 0 Black Black
% 6 2090 1800 2090 1600 2090 1400 2290 1250 2390 1400 2390 1400
% 
\special{pn 8}%
\special{pn 8}%
\special{pa 2090 1800}%
\special{pa 2092 1768}%
\special{pa 2093 1735}%
\special{fp}%
\special{pa 2094 1669}%
\special{pa 2094 1640}%
\special{pa 2092 1608}%
\special{pa 2091 1604}%
\special{fp}%
\special{pa 2081 1538}%
\special{pa 2078 1512}%
\special{pa 2078 1473}%
\special{fp}%
\special{pa 2089 1408}%
\special{pa 2098 1384}%
\special{pa 2116 1352}%
\special{pa 2118 1350}%
\special{fp}%
\special{pa 2162 1300}%
\special{pa 2166 1296}%
\special{pa 2196 1274}%
\special{pa 2215 1264}%
\special{fp}%
\special{pa 2279 1248}%
\special{pa 2282 1248}%
\special{pa 2306 1258}%
\special{pa 2328 1276}%
\special{pa 2332 1282}%
\special{fp}%
\special{pa 2365 1340}%
\special{pa 2390 1400}%
\special{fp}%
% SPLINE 2 1 3 0 Black Black
% 4 2490 1800 2490 1600 2490 1400 2690 1260
% 
\special{pn 8}%
\special{pn 8}%
\special{pa 2490 1800}%
\special{pa 2492 1770}%
\special{pa 2494 1741}%
\special{fp}%
\special{pa 2494 1682}%
\special{pa 2494 1642}%
\special{pa 2493 1623}%
\special{fp}%
\special{pa 2487 1563}%
\special{pa 2480 1508}%
\special{pa 2480 1505}%
\special{fp}%
\special{pa 2480 1445}%
\special{pa 2480 1444}%
\special{pa 2486 1414}%
\special{pa 2496 1389}%
\special{fp}%
\special{pa 2534 1344}%
\special{pa 2560 1324}%
\special{pa 2583 1311}%
\special{fp}%
\special{pa 2636 1284}%
\special{pa 2652 1276}%
\special{pa 2686 1262}%
\special{pa 2690 1260}%
\special{fp}%
% SPLINE 2 1 3 0 Black Black
% 4 3290 1260 3360 1280 3390 1400 3390 1400
% 
\special{pn 8}%
\special{pn 8}%
\special{pa 3290 1260}%
\special{pa 3324 1266}%
\special{pa 3354 1276}%
\special{pa 3354 1276}%
\special{fp}%
\special{pa 3386 1334}%
\special{pa 3390 1358}%
\special{pa 3390 1400}%
\special{fp}%
% VECTOR 0 0 3 0 Black Black
% 2 2290 2200 2290 2600
% 
\special{pn 20}%
\special{pa 2290 2200}%
\special{pa 2290 2600}%
\special{fp}%
\special{sh 1}%
\special{pa 2290 2600}%
\special{pa 2310 2534}%
\special{pa 2290 2548}%
\special{pa 2270 2534}%
\special{pa 2290 2600}%
\special{fp}%
% CIRCLE 2 0 0 0 Black Black
% 4 990 3192 990 3242 990 3242 990 3242
% 
\special{sh 1.000}%
\special{ia 990 3192 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 990 3192 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1090 3592 1090 3642 1090 3642 1090 3642
% 
\special{sh 1.000}%
\special{ia 1090 3592 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 1090 3592 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1190 3192 1190 3242 1190 3242 1190 3242
% 
\special{sh 1.000}%
\special{ia 1190 3192 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 1190 3192 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1390 3192 1390 3242 1390 3242 1390 3242
% 
\special{sh 1.000}%
\special{ia 1390 3192 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 1390 3192 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1590 3192 1590 3242 1590 3242 1590 3242
% 
\special{sh 1.000}%
\special{ia 1590 3192 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 1590 3192 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1290 3592 1290 3642 1290 3642 1290 3642
% 
\special{sh 1.000}%
\special{ia 1290 3592 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 1290 3592 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1490 3592 1490 3642 1490 3642 1490 3642
% 
\special{sh 1.000}%
\special{ia 1490 3592 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 1490 3592 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1790 3192 1790 3242 1790 3242 1790 3242
% 
\special{sh 1.000}%
\special{ia 1790 3192 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 1790 3192 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1690 3592 1690 3642 1690 3642 1690 3642
% 
\special{sh 1.000}%
\special{ia 1690 3592 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 1690 3592 50 50  0.0000000  6.2831853}%
% LINE 0 0 3 0 Black Black
% 2 990 3192 1090 3592
% 
\special{pn 20}%
\special{pa 990 3192}%
\special{pa 1090 3592}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1190 3192 1290 3592
% 
\special{pn 20}%
\special{pa 1190 3192}%
\special{pa 1290 3592}%
\special{fp}%
% LINE 2 1 3 0 Black Black
% 2 1390 3192 1490 3592
% 
\special{pn 8}%
\special{pa 1390 3192}%
\special{pa 1490 3592}%
\special{da 0.070}%
% LINE 0 0 3 0 Black Black
% 2 1590 3192 1690 3592
% 
\special{pn 20}%
\special{pa 1590 3192}%
\special{pa 1690 3592}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1790 3192 1690 3592
% 
\special{pn 20}%
\special{pa 1790 3192}%
\special{pa 1690 3592}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1590 3192 1490 3592
% 
\special{pn 20}%
\special{pa 1590 3192}%
\special{pa 1490 3592}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1390 3192 1290 3592
% 
\special{pn 20}%
\special{pa 1390 3192}%
\special{pa 1290 3592}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1190 3192 1090 3592
% 
\special{pn 20}%
\special{pa 1190 3192}%
\special{pa 1090 3592}%
\special{fp}%
% CIRCLE 2 0 0 0 Black Black
% 4 1985 3192 1985 3242 1985 3242 1985 3242
% 
\special{sh 1.000}%
\special{ia 1986 3192 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 1986 3192 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 2085 3592 2085 3642 2085 3642 2085 3642
% 
\special{sh 1.000}%
\special{ia 2086 3592 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 2086 3592 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 2185 3192 2185 3242 2185 3242 2185 3242
% 
\special{sh 1.000}%
\special{ia 2186 3192 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 2186 3192 50 50  0.0000000  6.2831853}%
% LINE 2 1 3 0 Black Black
% 2 1985 3192 2085 3592
% 
\special{pn 8}%
\special{pa 1986 3192}%
\special{pa 2086 3592}%
\special{da 0.070}%
% LINE 0 0 3 0 Black Black
% 2 2185 3192 2085 3592
% 
\special{pn 20}%
\special{pa 2186 3192}%
\special{pa 2086 3592}%
\special{fp}%
% DOT 0 0 3 0 Black Black
% 4 2995 3392 2895 3392 3095 3392 3095 3392
% 
\special{pn 4}%
\special{sh 1}%
\special{ar 2996 3392 16 16 0  6.28318530717959E+0000}%
\special{sh 1}%
\special{ar 2896 3392 16 16 0  6.28318530717959E+0000}%
\special{sh 1}%
\special{ar 3096 3392 16 16 0  6.28318530717959E+0000}%
\special{sh 1}%
\special{ar 3096 3392 16 16 0  6.28318530717959E+0000}%
% STR 2 0 3 0 Black Black
% 4 1195 3692 1195 3792 5 0 0 0
% $B_{i+1}$
\put(11.9500,-37.9200){\makebox(0,0){$B_{i+1}$}}%
% CIRCLE 2 0 0 0 Black Black
% 4 2385 3192 2385 3242 2385 3242 2385 3242
% 
\special{sh 1.000}%
\special{ia 2386 3192 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 2386 3192 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 2485 3592 2485 3642 2485 3642 2485 3642
% 
\special{sh 1.000}%
\special{ia 2486 3592 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 2486 3592 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 2585 3192 2585 3242 2585 3242 2585 3242
% 
\special{sh 1.000}%
\special{ia 2586 3192 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 2586 3192 50 50  0.0000000  6.2831853}%
% LINE 2 1 3 0 Black Black
% 2 2385 3192 2485 3592
% 
\special{pn 8}%
\special{pa 2386 3192}%
\special{pa 2486 3592}%
\special{da 0.070}%
% LINE 0 0 3 0 Black Black
% 2 2585 3192 2485 3592
% 
\special{pn 20}%
\special{pa 2586 3192}%
\special{pa 2486 3592}%
\special{fp}%
% CIRCLE 2 0 0 0 Black Black
% 4 3385 3192 3385 3242 3385 3242 3385 3242
% 
\special{sh 1.000}%
\special{ia 3386 3192 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 3386 3192 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3485 3592 3485 3642 3485 3642 3485 3642
% 
\special{sh 1.000}%
\special{ia 3486 3592 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 3486 3592 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3585 3192 3585 3242 3585 3242 3585 3242
% 
\special{sh 1.000}%
\special{ia 3586 3192 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 3586 3192 50 50  0.0000000  6.2831853}%
% LINE 2 1 3 0 Black Black
% 2 3385 3192 3485 3592
% 
\special{pn 8}%
\special{pa 3386 3192}%
\special{pa 3486 3592}%
\special{da 0.070}%
% LINE 0 0 3 0 Black Black
% 2 3585 3192 3485 3592
% 
\special{pn 20}%
\special{pa 3586 3192}%
\special{pa 3486 3592}%
\special{fp}%
% CIRCLE 2 0 0 0 Black Black
% 4 3785 3192 3785 3242 3785 3242 3785 3242
% 
\special{sh 1.000}%
\special{ia 3786 3192 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 3786 3192 50 50  0.0000000  6.2831853}%
% SPLINE 0 0 3 0 Black Black
% 6 1485 3592 1485 3392 1485 3192 1685 2992 1995 3192 1995 3192
% 
\special{pn 20}%
\special{pa 1486 3592}%
\special{pa 1488 3496}%
\special{pa 1488 3464}%
\special{pa 1486 3400}%
\special{pa 1478 3304}%
\special{pa 1476 3272}%
\special{pa 1478 3240}%
\special{pa 1482 3208}%
\special{pa 1490 3176}%
\special{pa 1504 3146}%
\special{pa 1522 3116}%
\special{pa 1542 3088}%
\special{pa 1566 3062}%
\special{pa 1592 3038}%
\special{pa 1618 3020}%
\special{pa 1648 3004}%
\special{pa 1676 2994}%
\special{pa 1704 2990}%
\special{pa 1732 2992}%
\special{pa 1760 2998}%
\special{pa 1788 3008}%
\special{pa 1814 3024}%
\special{pa 1842 3042}%
\special{pa 1894 3086}%
\special{pa 1920 3112}%
\special{pa 1972 3168}%
\special{pa 1996 3192}%
\special{fp}%
% SPLINE 0 0 3 0 Black Black
% 6 2085 3592 2085 3392 2085 3192 2285 3042 2385 3192 2385 3192
% 
\special{pn 20}%
\special{pa 2086 3592}%
\special{pa 2088 3560}%
\special{pa 2090 3496}%
\special{pa 2090 3464}%
\special{pa 2086 3400}%
\special{pa 2074 3304}%
\special{pa 2072 3272}%
\special{pa 2074 3240}%
\special{pa 2080 3208}%
\special{pa 2092 3176}%
\special{pa 2112 3144}%
\special{pa 2134 3114}%
\special{pa 2162 3088}%
\special{pa 2192 3066}%
\special{pa 2220 3050}%
\special{pa 2250 3042}%
\special{pa 2278 3040}%
\special{pa 2302 3050}%
\special{pa 2322 3068}%
\special{pa 2340 3094}%
\special{pa 2356 3124}%
\special{pa 2372 3158}%
\special{pa 2386 3192}%
\special{fp}%
% SPLINE 0 0 3 0 Black Black
% 4 2485 3592 2485 3392 2485 3192 2685 3052
% 
\special{pn 20}%
\special{pa 2486 3592}%
\special{pa 2488 3562}%
\special{pa 2488 3530}%
\special{pa 2490 3498}%
\special{pa 2490 3466}%
\special{pa 2488 3434}%
\special{pa 2486 3400}%
\special{pa 2482 3368}%
\special{pa 2478 3334}%
\special{pa 2474 3266}%
\special{pa 2476 3236}%
\special{pa 2480 3206}%
\special{pa 2492 3180}%
\special{pa 2508 3156}%
\special{pa 2530 3136}%
\special{pa 2554 3116}%
\special{pa 2614 3084}%
\special{pa 2648 3068}%
\special{pa 2682 3054}%
\special{pa 2686 3052}%
\special{fp}%
% SPLINE 0 0 3 0 Black Black
% 4 3285 3052 3355 3072 3385 3192 3385 3192
% 
\special{pn 20}%
\special{pa 3286 3052}%
\special{pa 3320 3058}%
\special{pa 3348 3068}%
\special{pa 3368 3088}%
\special{pa 3380 3116}%
\special{pa 3384 3150}%
\special{pa 3386 3188}%
\special{pa 3386 3192}%
\special{fp}%
% SPLINE 2 1 3 0 Black Black
% 6 3490 1800 3490 1600 3490 1400 3690 1250 3790 1400 3790 1400
% 
\special{pn 8}%
\special{pn 8}%
\special{pa 3490 1800}%
\special{pa 3492 1768}%
\special{pa 3493 1735}%
\special{fp}%
\special{pa 3494 1669}%
\special{pa 3494 1640}%
\special{pa 3492 1608}%
\special{pa 3491 1604}%
\special{fp}%
\special{pa 3481 1538}%
\special{pa 3478 1512}%
\special{pa 3478 1473}%
\special{fp}%
\special{pa 3489 1408}%
\special{pa 3498 1384}%
\special{pa 3516 1352}%
\special{pa 3518 1350}%
\special{fp}%
\special{pa 3562 1300}%
\special{pa 3566 1296}%
\special{pa 3596 1274}%
\special{pa 3615 1264}%
\special{fp}%
\special{pa 3679 1248}%
\special{pa 3682 1248}%
\special{pa 3706 1258}%
\special{pa 3728 1276}%
\special{pa 3732 1282}%
\special{fp}%
\special{pa 3765 1340}%
\special{pa 3790 1400}%
\special{fp}%
% SPLINE 0 0 3 0 Black Black
% 6 3490 3600 3490 3400 3490 3200 3690 3050 3790 3200 3790 3200
% 
\special{pn 20}%
\special{pa 3490 3600}%
\special{pa 3492 3568}%
\special{pa 3494 3504}%
\special{pa 3494 3440}%
\special{pa 3492 3408}%
\special{pa 3488 3376}%
\special{pa 3482 3344}%
\special{pa 3478 3312}%
\special{pa 3478 3248}%
\special{pa 3486 3216}%
\special{pa 3498 3184}%
\special{pa 3516 3152}%
\special{pa 3540 3122}%
\special{pa 3566 3096}%
\special{pa 3596 3074}%
\special{pa 3626 3058}%
\special{pa 3656 3050}%
\special{pa 3682 3048}%
\special{pa 3706 3058}%
\special{pa 3728 3076}%
\special{pa 3746 3102}%
\special{pa 3762 3132}%
\special{pa 3790 3200}%
\special{fp}%
% STR 2 0 3 0 Black Black
% 4 1720 2780 1720 2880 5 0 0 0
% $B_{i}$
\put(17.2000,-28.8000){\makebox(0,0){$B_{i}$}}%
% STR 2 0 3 0 Black Black
% 4 2220 2780 2220 2880 5 0 0 0
% $B_{i-1}$
\put(22.2000,-28.8000){\makebox(0,0){$B_{i-1}$}}%
% STR 2 0 3 0 Black Black
% 4 2690 2780 2690 2880 5 0 0 0
% $B_{i-2}$
\put(26.9000,-28.8000){\makebox(0,0){$B_{i-2}$}}%
% STR 2 0 3 0 Black Black
% 4 3620 2780 3620 2880 5 0 0 0
% $B_{1}$
\put(36.2000,-28.8000){\makebox(0,0){$B_{1}$}}%
\end{picture}}%

\caption{Construction of $B_{j}$}
\label{claim3.1fig1}
\end{center}
\end{figure}


We continue with the proof of the lemma.
Let $X_{0}=(\bigcup _{A\in \A}V(A))\cap S$ and $Y_{0}=((\bigcup _{A\in \A}V(A))\cap T)\cup (V(G)-V(F))$.
Since $V(G)-V(F)\not=\emptyset $ and $\A\subseteq \C_{3}(F)$ by Claim~\ref{cl2.1.1}, we have
\begin{align}
|Y_{0}\cap T_{1}|+\frac{2}{3}|Y_{0}\cap T_{2}|\geq \frac{2}{3}|Y_{0}|\geq \frac{2}{3}(2|X_{0}|+1).\label{lem2.1-1}
\end{align}
By the definition of $\A$, $N_{G}(S-X_{0})\cap Y_{0}=\emptyset $.
In particular, $N_{G}(S-X_{0})\not=T$, and hence $|N_{G}(S-X_{0})\cap T_{1}|+\frac{2}{3}|N_{G}(S-X_{0})\cap T_{2}|\geq \frac{4}{3}|S-X_{0}|$.
This together with (\ref{lem2.1-1}) implies that
\begin{eqnarray*}
|T_{1}|+\frac{2}{3}|T_{2}| &\geq & (|Y_{0}\cap T_{1}|+|N_{G}(S-X_{0})\cap T_{1}|)+\frac{2}{3}(|Y_{0}\cap T_{2}|+|N_{G}(S-X_{0})\cap T_{2}|)\\
&\geq & \frac{2}{3}(2|X_{0}|+1)+\frac{4}{3}|S-X_{0}|\\
&=& \frac{4}{3}|S|+\frac{2}{3},
\end{eqnarray*}
which contradicts the assumption that $|T_{1}|+\frac{2}{3}|T_{2}|\leq \frac{4}{3}|S|+\frac{1}{3}$, completing the proof of the lemma.
\qed

We here outline the proof of Theorem~\ref{thm2}.
We choose an $S$-central path-factor $F_{0}$ so that $F_{0}$ will satisfy certain minimality conditions (see the paragraph following the proof of Claim~\ref{cl2.1}).
We then introduce operations which turn $F_{0}$ into a new path-factor (see the paragraphs following Claim~\ref{cl-type2} and Claim~\ref{cl2.2}), and show that the new path-factor contradicts our choice of $F_{0}$.

\medbreak\noindent\textit{Proof of Theorem~\ref{thm2}.}\quad
We start with some definitions.
Let $F$ be an $S$-central path-factor of $G$.
For each integer $i\geq 2$, let $\C^{(1)}_{i}(F)=\{A\in \C_{i}(F)\mid V(A)\cap T_{2}=\emptyset \}$ and $\C^{(2)}_{i}(F)=\C_{i}(F)-\C^{(1)}_{i}(F)$.
If there is no fear of confusion, we simply write $\C_{i}$ and $\C^{(h)}_{i}~(h\in \{1,2\})$ instead of $\C_{i}(F)$ and $\C^{(h)}_{i}(F)$, respectively.

Let $\D_{F}$ be the digraph defined by $V(\D_{F})=\C(F)$ and $E(\D)=\{AB\mid E_{G}(V(A)\cap S,V(B)\cap T)\not=\emptyset \}$.
For each edge $AB\in E(\D_{F})$, we fix an edge $\varphi _{F}(AB)$ in $E_{G}(V(A)\cap S,V(B)\cap T)$, and let $\sigma _{F}(AB)\in V(G)$ be the vertex of $A$ incident with $\varphi _{F}(AB)$ and $\tau _{F}(AB)\in V(G)$ be the vertex of $B$ incident with $\varphi _{F}(AB)$ (see Figure~\ref{thm2.1fig1}).

\begin{figure}
\begin{center}
%WinTpicVersion4.26
\unitlength 0.1in
\begin{picture}( 24.0500, 11.0000)( 15.0500,-16.7500)
% CIRCLE 2 0 0 0 Black Black
% 4 1800 1000 1800 1050 1800 1050 1800 1050
% 
\special{sh 1.000}%
\special{ia 1800 1000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1800 1000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1900 1400 1900 1450 1900 1450 1900 1450
% 
\special{sh 1.000}%
\special{ia 1900 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1900 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 2000 1000 2000 1050 2000 1050 2000 1050
% 
\special{sh 1.000}%
\special{ia 2000 1000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 2000 1000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 2200 1000 2200 1050 2200 1050 2200 1050
% 
\special{sh 1.000}%
\special{ia 2200 1000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 2200 1000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 2400 1000 2400 1050 2400 1050 2400 1050
% 
\special{sh 1.000}%
\special{ia 2400 1000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 2400 1000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 2100 1400 2100 1450 2100 1450 2100 1450
% 
\special{sh 1.000}%
\special{ia 2100 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 2100 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 2300 1400 2300 1450 2300 1450 2300 1450
% 
\special{sh 1.000}%
\special{ia 2300 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 2300 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 2600 1000 2600 1050 2600 1050 2600 1050
% 
\special{sh 1.000}%
\special{ia 2600 1000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 2600 1000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 2500 1400 2500 1450 2500 1450 2500 1450
% 
\special{sh 1.000}%
\special{ia 2500 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 2500 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3200 1000 3200 1050 3200 1050 3200 1050
% 
\special{sh 1.000}%
\special{ia 3200 1000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3200 1000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3400 1000 3400 1050 3400 1050 3400 1050
% 
\special{sh 1.000}%
\special{ia 3400 1000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3400 1000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3600 1000 3600 1050 3600 1050 3600 1050
% 
\special{sh 1.000}%
\special{ia 3600 1000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3600 1000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3300 1400 3300 1450 3300 1450 3300 1450
% 
\special{sh 1.000}%
\special{ia 3300 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3300 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3500 1400 3500 1450 3500 1450 3500 1450
% 
\special{sh 1.000}%
\special{ia 3500 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3500 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3800 1000 3800 1050 3800 1050 3800 1050
% 
\special{sh 1.000}%
\special{ia 3800 1000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3800 1000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3700 1400 3700 1450 3700 1450 3700 1450
% 
\special{sh 1.000}%
\special{ia 3700 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3700 1400 50 50  0.0000000  6.2831853}%
% LINE 0 0 3 0 Black Black
% 2 1800 1000 1900 1400
% 
\special{pn 20}%
\special{pa 1800 1000}%
\special{pa 1900 1400}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 2000 1000 2100 1400
% 
\special{pn 20}%
\special{pa 2000 1000}%
\special{pa 2100 1400}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 2200 1000 2300 1400
% 
\special{pn 20}%
\special{pa 2200 1000}%
\special{pa 2300 1400}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 2400 1000 2500 1400
% 
\special{pn 20}%
\special{pa 2400 1000}%
\special{pa 2500 1400}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 3200 1000 3300 1400
% 
\special{pn 20}%
\special{pa 3200 1000}%
\special{pa 3300 1400}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 3400 1000 3500 1400
% 
\special{pn 20}%
\special{pa 3400 1000}%
\special{pa 3500 1400}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 3600 1000 3700 1400
% 
\special{pn 20}%
\special{pa 3600 1000}%
\special{pa 3700 1400}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 3600 1000 3500 1400
% 
\special{pn 20}%
\special{pa 3600 1000}%
\special{pa 3500 1400}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 3800 1000 3700 1400
% 
\special{pn 20}%
\special{pa 3800 1000}%
\special{pa 3700 1400}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 3400 1000 3300 1400
% 
\special{pn 20}%
\special{pa 3400 1000}%
\special{pa 3300 1400}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 2600 1000 2500 1400
% 
\special{pn 20}%
\special{pa 2600 1000}%
\special{pa 2500 1400}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 2400 1000 2300 1400
% 
\special{pn 20}%
\special{pa 2400 1000}%
\special{pa 2300 1400}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 2200 1000 2100 1400
% 
\special{pn 20}%
\special{pa 2200 1000}%
\special{pa 2100 1400}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 2000 1000 1900 1400
% 
\special{pn 20}%
\special{pa 2000 1000}%
\special{pa 1900 1400}%
\special{fp}%
% SPLINE 2 0 3 0 Black Black
% 6 2300 1400 2300 1000 2410 850 3210 850 3400 1000 3400 1000
% 
\special{pn 8}%
\special{pa 2300 1400}%
\special{pa 2294 1368}%
\special{pa 2288 1334}%
\special{pa 2284 1300}%
\special{pa 2280 1268}%
\special{pa 2276 1234}%
\special{pa 2272 1202}%
\special{pa 2272 1138}%
\special{pa 2274 1106}%
\special{pa 2280 1076}%
\special{pa 2286 1044}%
\special{pa 2294 1016}%
\special{pa 2306 986}%
\special{pa 2338 930}%
\special{pa 2358 904}%
\special{pa 2380 880}%
\special{pa 2402 858}%
\special{pa 2428 836}%
\special{pa 2454 818}%
\special{pa 2482 800}%
\special{pa 2510 786}%
\special{pa 2540 772}%
\special{pa 2572 762}%
\special{pa 2602 752}%
\special{pa 2636 744}%
\special{pa 2668 738}%
\special{pa 2702 734}%
\special{pa 2736 732}%
\special{pa 2772 730}%
\special{pa 2806 730}%
\special{pa 2840 732}%
\special{pa 2876 736}%
\special{pa 2910 742}%
\special{pa 2946 748}%
\special{pa 3014 764}%
\special{pa 3046 776}%
\special{pa 3080 786}%
\special{pa 3112 800}%
\special{pa 3172 828}%
\special{pa 3202 846}%
\special{pa 3230 862}%
\special{pa 3256 880}%
\special{pa 3308 920}%
\special{pa 3356 960}%
\special{pa 3380 982}%
\special{pa 3400 1000}%
\special{sp}%
% STR 2 0 3 0 Black Black
% 4 1600 1300 1600 1400 5 0 0 0
% $A$
\put(16.0000,-14.0000){\makebox(0,0){$A$}}%
% STR 2 0 3 0 Black Black
% 4 4000 1300 4000 1400 5 0 0 0
% $B$
\put(40.0000,-14.0000){\makebox(0,0){$B$}}%
% STR 2 0 3 0 Black Black
% 4 2300 1640 2300 1740 5 0 0 0
% $\sigma _{F}(AB)$
\put(23.0000,-17.4000){\makebox(0,0){$\sigma _{F}(AB)$}}%
% STR 2 0 3 0 Black Black
% 4 3660 620 3660 720 5 0 0 0
% $\tau _{F}(AB)$
\put(36.6000,-7.2000){\makebox(0,0){$\tau _{F}(AB)$}}%
% STR 2 0 3 0 Black Black
% 4 2600 540 2600 640 5 0 0 0
% $\varphi _{F}(AB)$
\put(26.0000,-6.4000){\makebox(0,0){$\varphi _{F}(AB)$}}%
% VECTOR 2 0 3 0 Black Black
% 2 2300 1630 2300 1480
% 
\special{pn 8}%
\special{pa 2300 1630}%
\special{pa 2300 1480}%
\special{fp}%
\special{sh 1}%
\special{pa 2300 1480}%
\special{pa 2280 1548}%
\special{pa 2300 1534}%
\special{pa 2320 1548}%
\special{pa 2300 1480}%
\special{fp}%
% VECTOR 2 0 3 0 Black Black
% 2 3600 800 3450 940
% 
\special{pn 8}%
\special{pa 3600 800}%
\special{pa 3450 940}%
\special{fp}%
\special{sh 1}%
\special{pa 3450 940}%
\special{pa 3512 910}%
\special{pa 3490 904}%
\special{pa 3486 880}%
\special{pa 3450 940}%
\special{fp}%
\end{picture}%

\caption{Edge $\varphi _{F}(AB)$ and vertices $\sigma _{F}(AB)$ and $\tau _{F}(AB)$}
\label{thm2.1fig1}
\end{center}
\end{figure}

For a path $A=x_{1}x_{2}\cdots x_{7}\in \C_{7}$, the vertex $x_{4}$ is called the {\it center} of $A$.
A directed path $\P=A_{1}A_{2}\cdots A_{l}~(l\geq 2)$ of $\D_{F}$ is {\it admissible} if $A_{1}\in \C(F)-(\C_{3}\cup \C^{(1)}_{5})$ and $A_{i}\in \C^{(2)}_{3}\cup \C^{(1)}_{5}$ for every $i~(2\leq i\leq l-1)$.
An admissible path $\P=A_{1}A_{2}\cdots A_{l}$ of $\D_{F}$ is {\it weakly admissible} if either
\begin{enumerate}
\item[{\bf (W1)}]
$A_{1}\in \C^{(2)}_{5}$ and $|V(A_{1})\cap T_{2}|=1$, or
\item[{\bf (W2)}]
$A_{1}\in \C^{(1)}_{7}$ and $\sigma _{F}(A_{1}A_{2})$ is the center of $A_{1}$.
\end{enumerate} 
An admissible path $\P$ of $\D_{F}$ is {\it strongly admissible} if $\P$ is not weakly admissible.

A {\it path system} with respect to $F$ is a sequence $(\P_{1},\ldots,\P_{m})~(m\geq 0)$ of admissible paths such that 
\begin{enumerate}
\item[{\bf (P1)}]
for each $i~(1\leq i\leq m)$, when we write $\P_{i}=A_{1}A_{2}\cdots A_{l}$, $\{A_{j}\mid 1\leq j\leq l-1\}\cap (\bigcup _{1\leq j\leq i-1}V(\P_{j}))=\emptyset $ and $A_{l}\in \C^{(1)}_{3}\cup (\bigcup _{1\leq j\leq i-1}V(\P_{j}))$, and
\item[{\bf (P2)}]
for each $i~(1\leq i\leq m-1)$, $\P_{i}$ is weakly admissible.
\end{enumerate}
A path system $(\P_{1},\ldots,\P_{m})$ with respect to $F$ is {\it complete} if $m\geq 1$ and $\P_{m}$ is strongly admissible.

By straightforward calculations, we get the following claim (and we omit its proof).

\begin{claim}%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\label{cl2.0}
Let $F$ be an $S$-central path-factor of $G$.
Then the following hold.
\begin{enumerate}
\item[{\upshape(i)}]
For $A\in \C^{(1)}_{3}(F)$, $|V(A)\cap T_{1}|+\frac{2}{3}|V(A)\cap T_{2}|=2=\frac{4}{3}|V(A)\cap S|+\frac{2}{3}$.
\item[{\upshape(ii)}]
For $A\in \C^{(2)}_{3}(F)$, $|V(A)\cap T_{1}|+\frac{2}{3}|V(A)\cap T_{2}|\geq \frac{4}{3}|V(A)\cap S|$.
\item[{\upshape(iii)}]
For $A\in \C^{(1)}_{5}(F)$, $|V(A)\cap T_{1}|+\frac{2}{3}|V(A)\cap T_{2}|>\frac{4}{3}|V(A)\cap S|$.
\item[{\upshape(iv)}]
For $A\in \C^{(2)}_{5}(F)$ with $|V(A)\cap T_{2}|=1$, $|V(A)\cap T_{1}|+\frac{2}{3}|V(A)\cap T_{2}|=\frac{4}{3}|V(A)\cap S|$.
\item[{\upshape(v)}]
For $A\in \C^{(1)}_{7}(F)$, $|V(A)\cap T_{1}|+\frac{2}{3}|V(A)\cap T_{2}|=\frac{4}{3}|V(A)\cap S|$.
\qed
\end{enumerate}
\end{claim}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

The following claim plays a key role in the proof of the theorem.

\begin{claim}%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\label{cl2.1}
Let $F$ be an $S$-central path-factor of $G$ with $\C^{(1)}_{3}(F)\not=\emptyset $, and let $(\P_{1},\ldots,\P_{m})$ be a path system with respect to $F$ ($m\geq 0$).
Then the system can be extend to a complete path system $(\P_{1},\ldots,\P_{m},\P_{m+1},\ldots,\P_{m'})$ with respect to $F$.
\end{claim}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\proof
We take a maximal path system $(\P_{1},\ldots,\P_{m},\P_{m+1},\ldots,\P_{m'})$ with respect to $F$.
We show that $(\P_{1},\ldots,\P_{m'})$ is a complete path system.
Suppose that $(\P_{1},\ldots,\P_{m'})$ is not a complete path system.
Then $\P_{i}$ is weakly admissible for each $i$ with $1\leq i\leq m'$ (this includes the case where $m'=0$).

Set $\A_{1}=\bigcup _{1\leq i\leq m'}V(\P_{i})$ (note that $\A_{1}=\emptyset $ if and only if $m'=0$).
Let $X=(\bigcup _{A\in \A_{1}}V(A))\cap S$ and $Y_{h}=(\bigcup _{A\in \A_{1}}V(A))\cap T_{h}~(h\in \{1,2\})$.
Then by the definition of a weakly admissible path (and the definition of a path system), $\A_{1}\subseteq \C_{3}\cup \C_{5}\cup \C^{(1)}_{7}$, and if $A\in \A_{1}\cap \C^{(2)}_{5}$, then $|V(A)\cap T_{1}|=1$.
Furthermore, by condition (P1) in the definition of a path system, $\A_{1}\not=\emptyset $ if and only if $\A_{1}\cap \C^{(1)}_{3}\not=\emptyset $.
Hence by Claim~\ref{cl2.0},
\begin{align}
|Y_{1}|+\frac{2}{3}|Y_{2}|\geq \frac{4}{3}|X|\label{cl2.1-6}
\end{align}
and
\begin{align}
|Y_{1}|+\frac{2}{3}|Y_{2}|\geq \frac{4}{3}|X|+\frac{2}{3} \mbox{ if } \A_{1}\not=\emptyset.\label{cl2.1-7}
\end{align}

Let $\A_{2}=\C^{(1)}_{3}-\A_{1}$, $X^{*}=(\bigcup _{A\in \A_{2}}V(A))\cap S$ and $Y^{*}_{h}=(\bigcup _{A\in \A_{2}}V(A))\cap T_{h}~(h\in \{1,2\})$.
By Claim~\ref{cl2.0}(i),
\begin{align}
|Y^{*}_{1}|+\frac{2}{3}|Y^{*}_{2}|\geq \frac{4}{3}|X^{*}|\label{cl2.1-8}
\end{align}
and
\begin{align}
|Y^{*}_{1}|+\frac{2}{3}|Y^{*}_{2}|=\frac{4}{3}|X^{*}|+\frac{2}{3} \mbox{ if } \A_{2}\not=\emptyset,\label{cl2.1-9}
\end{align}

Let $(B_{1},\ldots,B_{l})~(l\geq 0)$ be a sequence such that for each $i~(1\leq i\leq l)$, $B_{i}\in (\C^{(2)}_{3}\cup \C^{(1)}_{5})-(\A_{1}\cup \{B_{j}\mid 1\leq j\leq i-1\})$ and there exists an edge of $\D_{F}$ from $B_{i}$ to an element in $\A_{1}\cup \A_{2}\cup \{B_{j}\mid 1\leq j\leq i-1\}$.
We choose $(B_{1},\ldots,B_{l})$ so that $l$ is as large as possible.
Let $\A_{3}=\{B_{i}\mid 1\leq i\leq l\}$, $X^{**}=(\bigcup _{A\in \A_{3}}V(A))\cap S$ and $Y^{**}_{h}=(\bigcup _{A\in \A_{3}}V(A))\cap T_{h}~(h\in \{1,2\})$.
By Claim~\ref{cl2.0}(ii)(iii),
\begin{align}
|Y^{**}_{1}|+\frac{2}{3}|Y^{**}_{2}|\geq \frac{4}{3}|X^{**}|.\label{cl2.1-10}
\end{align}

Let $X^{0}=X\cup X^{*}\cup X^{**}$ and $Y^{0}_{h}=Y_{h}\cup Y^{*}_{h}\cup Y^{**}_{h}~(h\in \{1,2\})$.
If $m'\geq 1$, then $\A_{1}\not=\emptyset $; if $m'=0$ (i.e., $\A_{1}=\emptyset $), then $\A_{2}\not=\emptyset $ because $\C^{(1)}_{3}\not=\emptyset $.
Thus by (\ref{cl2.1-7}) and (\ref{cl2.1-9}), either $|Y_{1}|+\frac{2}{3}|Y_{2}|\geq \frac{4}{3}|X|+\frac{2}{3}$ or $|Y^{*}_{1}|+\frac{2}{3}|Y^{*}_{2}|=\frac{4}{3}|X^{*}|+\frac{2}{3}$.
This together with (\ref{cl2.1-6}), (\ref{cl2.1-8}) and (\ref{cl2.1-10}) leads to
\begin{align}
|Y^{0}_{1}|+\frac{2}{3}|Y^{0}_{2}|\geq \frac{4}{3}|X^{0}|+\frac{2}{3}.\label{cl2.1-11}
\end{align}
Since $|T_{1}|+\frac{2}{3}|T_{2}|\leq \frac{4}{3}|S|+\frac{1}{3}$, this implies $X^{0}\not=S$ and hence $\C(F)-(\A_{1}\cup \A_{2}\cup \A_{3})\not=\emptyset $.

Let $\tilde{\A}=\C(F)-(\A_{1}\cup \A_{2}\cup \A_{3})$, $\tilde{X}=(\bigcup _{A\in \tilde{\A}}V(A))\cap S$ and $\tilde{Y}_{h}=(\bigcup _{A\in \tilde{\A}}V(A))\cap T_{h}~(h\in \{1,2\})$.
Note that $S$ is the disjoint union of $X^{0}$ and $\tilde{X}$ and, for $h\in \{1,2\}$, $T_{h}$ is the disjoint union of $Y^{0}_{h}$ and $\tilde{Y}_{h}$.
If $|\tilde{Y}_{1}|+\frac{2}{3}|\tilde{Y}_{2}|\geq \frac{4}{3}|\tilde{X}|$, then by (\ref{cl2.1-11}), $|T_{1}|+\frac{2}{3}|T_{2}|=(|Y^{0}_{1}|+|\tilde{Y}_{1}|)+\frac{2}{3}(|Y^{0}_{2}|+|\tilde{Y}_{2}|)\geq \frac{4}{3}|X^{0}|+\frac{2}{3}+\frac{4}{3}|\tilde{X}|=\frac{4}{3}|S|+\frac{2}{3}$, which is a contradiction.
Thus $|\tilde{Y}_{1}|+\frac{2}{3}|\tilde{Y}_{2}|<\frac{4}{3}|\tilde{X}|$.
On the other hand, since $\A_{1}\cup \A_{2}\not=\emptyset $, we have $Y^{0}_{1}\cup Y^{0}_{2}\not=\emptyset $, and hence $\tilde{Y}_{1}\cup \tilde{Y}_{2}\not=T$.
Consequently $N_{G}(\tilde{X})\not\subseteq \tilde{Y}_{1}\cup \tilde{Y}_{2}$ by the assumption of the theorem, which implies that there exists a vertex $x\in \tilde{X}$ with $N_{G}(x)\cap (Y^{0}_{1}\cup Y^{0}_{2})\not=\emptyset $.
Let $\tilde{A}\in \tilde{\A}$ be the path containing $x$.
By the definition of $\A_{2}$ and $\tilde{\A}$, $\tilde{A}\not\in \C^{(1)}_{3}$.
By the maximality of $(B_{1},\ldots,B_{l})$, $\tilde{A}\not\in \C^{(2)}_{3}\cup \C^{(1)}_{5}$.
Thus $\tilde{A}\in \C(F)-(\C_{3}\cup \C^{(1)}_{5})$.
By the definition of $(B_{1},\ldots,B_{l})$ and $x$, there exists a directed path $\P'=\tilde{A}_{1}\cdots \tilde{A}_{p}$ of $\D_{F}$ such that $\tilde{A}_{1}=\tilde{A}$, $\tilde{A}_{i}\in \A_{3}~(2\leq i\leq p-1)$ and $\tilde{A}_{p}\in \A_{1}\cup \A_{2}$.
Then $\P'$ is an admissible path of $\D_{F}$.
Now the sequence $(\P_{1},\ldots,\P_{m'},\P')$ is a path system with respect to $F$, which contradicts the maximality of $(\P_{1},\ldots,\P_{m'})$.
This contradiction completes the proof of the claim.
\qed

We turn to the proof of Theorem~\ref{thm2}.
By way of contradiction, suppose that $\C^{(1)}_{3}(F)\not=\emptyset $ for every $S$-central path-factor $F$ of $G$.
By Lemma~\ref{lem2.1}, $G$ has an $S$-central path-factor $F_{0}$.
Note that an empty sequence is a path system with respect to $F_{0}$.
Hence by Claim~\ref{cl2.1}, there exists a complete path system $(\P_{1},\ldots,\P_{m})$ with respect to $F_{0}$.
Choose $F_{0}$ and $(\P_{1},\ldots,\P_{m})$ so that
\begin{enumerate}
\item[{\bf (F1)}]
$|\C^{(1)}_{3}(F_{0})|$ is as small as possible, and
\item[{\bf (F2)}]
subject to (F1), $(|V(\P_{1})|,\ldots,|V(\P_{m})|)$ is lexicographically as small as possible.
\end{enumerate}

For each $i~(1\leq i\leq m)$, write $\P_{i}=A^{(i)}_{1}\cdots A^{(i)}_{l_{i}}$.
Then $\bigcup _{1\leq i\leq m}\P_{i}$ contains a directed path $B_{1}B_{2}\cdots B_{p}$ of $\D_{F_{0}}$ with $B_{1}=A^{(m)}_{1}$ and $B_{p}\in \C^{(1)}_{3}(F_{0})$.
For each $i~(1\leq i\leq p)$, write $B_{i}=v_{i,1}v_{i,2}\cdots v_{i,q_{i}}$.
For $i~(1\leq i\leq p-1)$, let $s_{i}$ be the integer with $v_{i,s_{i}}=\sigma _{F}(B_{i}B_{i+1})$, and for $i~(2\leq i\leq p)$, let $t_{i}$ be the integer with $v_{i,t_{i}}=\tau _{F_{0}}(B_{i-1}B_{i})$.
As in the proof of Claim~\ref{cl2.1.1}, by renumbering the vertices of some of the $B_{i}$ backward if necessary, we may assume that
\begin{enumerate}
\item[{\bf (B1)}]
$s_{1}\geq \frac{q_{1}+1}{2}$ if $q_{1}$ is odd,
\item[{\bf (B2)}]
$\{v_{1,1},v_{1,3}\}\cap T_{2}\not=\emptyset $ if $B_{1}\in \C^{(2)}_{7}(F_{0})$ and $s_{1}=4$,
\item[{\bf (B3)}]
$s_{1}$ is odd if $q_{1}$ is even,
\item[{\bf (B4)}]
$t_{i}<s_{i}$ for each $i~(2\leq i\leq p-1)$, and
\item[{\bf (B5)}]
$t_{p}=q_{p}~(=3)$.
\end{enumerate}
Note that (B3) means that when $q_{1}$ is even, the vertices of $B_{1}$ are numbered so that $v_{1,q_{1}}\in T$.
Thus $v_{i,q_{i}}\in T$ for each $i~(1\leq i\leq p)$.
We can divide the type of $B_{1}$ into three possibilities as follows:

\begin{claim}%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\label{cl-type1}
One of the following holds:
\begin{enumerate}
\item[{\upshape(1)}]
$|V(B_{1})|$ is even and $s_{1}$ is odd;
\item[{\upshape(2)}]
$B_{1}\in \C^{(2)}_{5}(F_{0})\cup \C^{(2)}_{7}(F_{0})$, $s_{1}=4$ and $\{v_{1,1},v_{1,3}\}\cap T_{2}\not=\emptyset $; or
\item[{\upshape(3)}]
$|V(B_{1})|\geq 7$ and $s_{1}\geq 6$.
\end{enumerate}
\end{claim}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\proof
If $|V(B_{1})|$ is even, then (1) holds by (B3).
Thus we may assume $|V(B_{1})|$ is odd.
Then by the definition of a strongly admissible path, $B_{1}\in \C^{(2)}_{5}(F_{0})$ and $|V(B_{1})\cap T_{2}|\geq 2$, or $B_{1}\in \C^{(1)}_{7}(F_{0})$ and $s_{1}\not=4$, or $B_{1}\in \C^{(2)}_{7}(F_{0})$, or $|V(B_{1})|\geq 9$.
If $B_{1}\in \C^{(2)}_{5}(F_{0})$ and $|V(B_{1})\cap T_{2}|\geq 2$, then (2) holds by (B1).
If $B_{1}\in \C^{(1)}_{7}(F_{0})\cup \C^{(2)}_{7}(F_{0})$ and $s_{1}\not=4$, then (3) holds by (B1).
If $B_{1}\in \C^{(2)}_{7}(F_{0})$ and $s_{1}=4$, then (2) holds by (B2).
If $|V(B_{1})|\geq 9$, then (3) holds by (B1).
\qed

As for $B_{i}$ with $2\leq i\leq p-1$, the following claim follows immediately from the definition of a weakly admissible path.

\begin{claim}%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\label{cl-type2}
Let $2\leq i\leq p-1$.
Then one of the following holds:
\begin{enumerate}
\item[{\upshape(1)}]
$B_{i}\in \C^{(2)}_{3}(F_{0})$ and $s_{i}=2$;
\item[{\upshape(2)}]
$B_{i}\in \C_{5}(F_{0})$ and $s_{i}=2$ or $4$; or
\item[{\upshape(3)}]
$B_{i}\in \C^{(1)}_{7}(F_{0})$ and $s_{i}=4$.
\qed
\end{enumerate}
\end{claim}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

Let $i_{0}$ be the minimum integer $i~(\geq 2)$ satisfying one of the following two conditions:
\begin{enumerate}
\item[{\bf (I1)}]
$i=p$; or
\item[{\bf (I2)}]
$2\leq i\leq p-1$ and $t_{i}=1$.
\end{enumerate}
Set $B'_{1}=v_{1,1}v_{1,2}\cdots v_{1,s_{1}-1}$ and, for each $i~(2\leq i\leq i_{0})$, set
$$
B'_{i}=v_{i-1,q_{i-1}}v_{i-1,q_{i-1}-1}\cdots v_{i-1,s_{i-1}}v_{i,t_{i}}v_{i,t_{i}-1}\cdots v_{i,1}
$$
(see Figure~\ref{thm2.1fig2}).
Let $2\leq i\leq i_{0}-1$.
By the definition of $i_{0}$, $t_{i}\geq 3$.
On the other hand, $s_{i}\leq 4$ by Claim~\ref{cl-type2}.
Hence it follows from (B4) that $t_{i}=s_{i}-1$.
Since $i~(2\leq i\leq i_{0}-1)$ is arbitrary, it follows that
\begin{align}
\label{neweq2.1}
B'_{1},\ldots,B'_{i_{0}}\mbox{ are vertex-disjoint paths of }G
\end{align}
and
\begin{align}
\label{neweq2.2}
\bigcup _{1\leq i\leq i_{0}}V(B'_{i})=\bigcup _{1\leq i\leq i_{0}}V(B_{i})-\{v_{i_{0},j}\mid t_{i_{0}}+1\leq j\leq q_{i_{0}}\}.
\end{align}
Furthermore, 
\begin{align}
\label{neweq2.3}
|V(B'_{i})\cap T|\geq |V(B'_{i})\cap S|\mbox{ for each }i~(2\leq i\leq i_{0})
\end{align}
because $v_{i-1,q_{i-1}}\in T$.
If $B'_{1}\not=\emptyset $, then $v_{1,s_{1}-1}\in T$, and hence
\begin{align}
\label{neweq2.4}
|V(B'_{1})\cap T|\geq |V(B'_{1})\cap S|
\end{align}
(if $B'_{1}=\emptyset $, then (\ref{neweq2.4}) trivially holds).
Also
\begin{align}
\label{neweq2.5}
|V(B'_{i})\cap V(B_{i-1})|\mbox{ is even and }|V(B'_{i})\cap V(B_{i-1})|\geq 2\mbox{ for each }i~(2\leq i\leq i_{0})
\end{align}
because $v_{i-1,s_{i-1}}\in S$ and $v_{i-1,q_{i-1}}\in T$.
It follows from (\ref{neweq2.5}) that
\begin{align}
\label{neweq2.6}
|V(B'_{i})|\geq 5\mbox{ for each }i~(2\leq i\leq i_{0}-1)
\end{align}
because $|V(B'_{i})\cap V(B_{i})|=t_{i}\geq 3$.
Since $|V(B'_{1})|=s_{1}-1$, we see from Claim~\ref{cl-type1} that
\begin{align}
\label{neweq2.7}
|V(B'_{1})|\mbox{ is even or }|V(B'_{1})|\geq 3,
\end{align}
and
\begin{align}
\label{neweq2.8}
V(B'_{1})\cap T_{2}\not=\emptyset \mbox{ if }|V(B'_{1})|=3.
\end{align}
Combining (\ref{neweq2.3}) through (\ref{neweq2.8}), we get the following claim.

\begin{figure}
\begin{center}
%WinTpicVersion4.26
\unitlength 0.1in
\begin{picture}( 44.8500, 22.8200)(  5.8500,-34.5000)
% CIRCLE 2 0 0 0 Black Black
% 4 995 1400 995 1450 995 1450 995 1450
% 
\special{sh 1.000}%
\special{ia 996 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 996 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1095 1800 1095 1850 1095 1850 1095 1850
% 
\special{sh 1.000}%
\special{ia 1096 1800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1096 1800 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1195 1400 1195 1450 1195 1450 1195 1450
% 
\special{sh 1.000}%
\special{ia 1196 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1196 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1395 1400 1395 1450 1395 1450 1395 1450
% 
\special{sh 1.000}%
\special{ia 1396 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1396 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1595 1400 1595 1450 1595 1450 1595 1450
% 
\special{sh 1.000}%
\special{ia 1596 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1596 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1295 1800 1295 1850 1295 1850 1295 1850
% 
\special{sh 1.000}%
\special{ia 1296 1800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1296 1800 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1495 1800 1495 1850 1495 1850 1495 1850
% 
\special{sh 1.000}%
\special{ia 1496 1800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1496 1800 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1795 1400 1795 1450 1795 1450 1795 1450
% 
\special{sh 1.000}%
\special{ia 1796 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1796 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1695 1800 1695 1850 1695 1850 1695 1850
% 
\special{sh 1.000}%
\special{ia 1696 1800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1696 1800 50 50  0.0000000  6.2831853}%
% LINE 0 0 3 0 Black Black
% 2 995 1400 1095 1800
% 
\special{pn 20}%
\special{pa 996 1400}%
\special{pa 1096 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1195 1400 1295 1800
% 
\special{pn 20}%
\special{pa 1196 1400}%
\special{pa 1296 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1395 1400 1495 1800
% 
\special{pn 20}%
\special{pa 1396 1400}%
\special{pa 1496 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1595 1400 1695 1800
% 
\special{pn 20}%
\special{pa 1596 1400}%
\special{pa 1696 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1795 1400 1695 1800
% 
\special{pn 20}%
\special{pa 1796 1400}%
\special{pa 1696 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1595 1400 1495 1800
% 
\special{pn 20}%
\special{pa 1596 1400}%
\special{pa 1496 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1395 1400 1295 1800
% 
\special{pn 20}%
\special{pa 1396 1400}%
\special{pa 1296 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1195 1400 1095 1800
% 
\special{pn 20}%
\special{pa 1196 1400}%
\special{pa 1096 1800}%
\special{fp}%
% CIRCLE 2 0 0 0 Black Black
% 4 2195 1400 2195 1450 2195 1450 2195 1450
% 
\special{sh 1.000}%
\special{ia 2196 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 2196 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 2295 1800 2295 1850 2295 1850 2295 1850
% 
\special{sh 1.000}%
\special{ia 2296 1800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 2296 1800 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 2395 1400 2395 1450 2395 1450 2395 1450
% 
\special{sh 1.000}%
\special{ia 2396 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 2396 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 2595 1400 2595 1450 2595 1450 2595 1450
% 
\special{sh 1.000}%
\special{ia 2596 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 2596 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 2495 1800 2495 1850 2495 1850 2495 1850
% 
\special{sh 1.000}%
\special{ia 2496 1800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 2496 1800 50 50  0.0000000  6.2831853}%
% LINE 0 0 3 0 Black Black
% 2 2195 1400 2295 1800
% 
\special{pn 20}%
\special{pa 2196 1400}%
\special{pa 2296 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 2395 1400 2495 1800
% 
\special{pn 20}%
\special{pa 2396 1400}%
\special{pa 2496 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 2595 1400 2495 1800
% 
\special{pn 20}%
\special{pa 2596 1400}%
\special{pa 2496 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 2395 1400 2295 1800
% 
\special{pn 20}%
\special{pa 2396 1400}%
\special{pa 2296 1800}%
\special{fp}%
% CIRCLE 2 0 0 0 Black Black
% 4 3395 1400 3395 1450 3395 1450 3395 1450
% 
\special{sh 1.000}%
\special{ia 3396 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3396 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3495 1800 3495 1850 3495 1850 3495 1850
% 
\special{sh 1.000}%
\special{ia 3496 1800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3496 1800 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3595 1400 3595 1450 3595 1450 3595 1450
% 
\special{sh 1.000}%
\special{ia 3596 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3596 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3795 1400 3795 1450 3795 1450 3795 1450
% 
\special{sh 1.000}%
\special{ia 3796 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3796 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3995 1400 3995 1450 3995 1450 3995 1450
% 
\special{sh 1.000}%
\special{ia 3996 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3996 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3695 1800 3695 1850 3695 1850 3695 1850
% 
\special{sh 1.000}%
\special{ia 3696 1800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3696 1800 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3895 1800 3895 1850 3895 1850 3895 1850
% 
\special{sh 1.000}%
\special{ia 3896 1800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3896 1800 50 50  0.0000000  6.2831853}%
% LINE 0 0 3 0 Black Black
% 2 3395 1400 3495 1800
% 
\special{pn 20}%
\special{pa 3396 1400}%
\special{pa 3496 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 3595 1400 3695 1800
% 
\special{pn 20}%
\special{pa 3596 1400}%
\special{pa 3696 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 3795 1400 3895 1800
% 
\special{pn 20}%
\special{pa 3796 1400}%
\special{pa 3896 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 3995 1400 3895 1800
% 
\special{pn 20}%
\special{pa 3996 1400}%
\special{pa 3896 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 3795 1400 3695 1800
% 
\special{pn 20}%
\special{pa 3796 1400}%
\special{pa 3696 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 3595 1400 3495 1800
% 
\special{pn 20}%
\special{pa 3596 1400}%
\special{pa 3496 1800}%
\special{fp}%
% CIRCLE 2 0 0 0 Black Black
% 4 4395 1400 4395 1450 4395 1450 4395 1450
% 
\special{sh 1.000}%
\special{ia 4396 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 4396 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 4495 1800 4495 1850 4495 1850 4495 1850
% 
\special{sh 1.000}%
\special{ia 4496 1800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 4496 1800 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 4595 1400 4595 1450 4595 1450 4595 1450
% 
\special{sh 1.000}%
\special{ia 4596 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 4596 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 4795 1400 4795 1450 4795 1450 4795 1450
% 
\special{sh 1.000}%
\special{ia 4796 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 4796 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 4695 1800 4695 1850 4695 1850 4695 1850
% 
\special{sh 1.000}%
\special{ia 4696 1800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 4696 1800 50 50  0.0000000  6.2831853}%
% LINE 0 0 3 0 Black Black
% 2 4395 1400 4495 1800
% 
\special{pn 20}%
\special{pa 4396 1400}%
\special{pa 4496 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 4595 1400 4695 1800
% 
\special{pn 20}%
\special{pa 4596 1400}%
\special{pa 4696 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 4795 1400 4695 1800
% 
\special{pn 20}%
\special{pa 4796 1400}%
\special{pa 4696 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 4595 1400 4495 1800
% 
\special{pn 20}%
\special{pa 4596 1400}%
\special{pa 4496 1800}%
\special{fp}%
% SPLINE 2 1 3 0 Black Black
% 4 3200 1200 3400 1260 3600 1400 3600 1400
% 
\special{pn 8}%
\special{pn 8}%
\special{pa 3200 1200}%
\special{pa 3262 1216}%
\special{fp}%
\special{pa 3326 1232}%
\special{pa 3326 1232}%
\special{pa 3356 1242}%
\special{pa 3386 1254}%
\special{pa 3386 1254}%
\special{fp}%
\special{pa 3446 1284}%
\special{pa 3468 1300}%
\special{pa 3496 1318}%
\special{pa 3498 1320}%
\special{fp}%
\special{pa 3550 1360}%
\special{pa 3572 1376}%
\special{pa 3596 1398}%
\special{pa 3600 1400}%
\special{fp}%
% SPLINE 2 1 3 0 Black Black
% 6 3700 1800 3700 1400 3800 1280 4200 1280 4400 1400 4400 1400
% 
\special{pn 8}%
\special{pn 8}%
\special{pa 3700 1800}%
\special{pa 3694 1766}%
\special{pa 3690 1738}%
\special{fp}%
\special{pa 3678 1674}%
\special{pa 3676 1666}%
\special{pa 3672 1634}%
\special{pa 3672 1612}%
\special{fp}%
\special{pa 3668 1548}%
\special{pa 3668 1536}%
\special{pa 3672 1506}%
\special{pa 3676 1484}%
\special{fp}%
\special{pa 3692 1422}%
\special{pa 3694 1414}%
\special{pa 3708 1386}%
\special{pa 3720 1366}%
\special{fp}%
\special{pa 3760 1316}%
\special{pa 3766 1308}%
\special{pa 3790 1288}%
\special{pa 3810 1276}%
\special{fp}%
\special{pa 3868 1250}%
\special{pa 3876 1246}%
\special{pa 3908 1238}%
\special{pa 3930 1236}%
\special{fp}%
\special{pa 3994 1232}%
\special{pa 4006 1232}%
\special{pa 4040 1236}%
\special{pa 4056 1240}%
\special{fp}%
\special{pa 4120 1252}%
\special{pa 4140 1258}%
\special{pa 4180 1274}%
\special{fp}%
\special{pa 4238 1300}%
\special{pa 4294 1330}%
\special{fp}%
\special{pa 4348 1366}%
\special{pa 4396 1398}%
\special{pa 4400 1400}%
\special{fp}%
% SPLINE 2 1 3 0 Black Black
% 5 4700 1800 4700 1400 4800 1280 5070 1200 5070 1200
% 
\special{pn 8}%
\special{pn 8}%
\special{pa 4700 1800}%
\special{pa 4694 1766}%
\special{pa 4688 1736}%
\special{fp}%
\special{pa 4678 1672}%
\special{pa 4676 1666}%
\special{pa 4672 1634}%
\special{pa 4670 1608}%
\special{fp}%
\special{pa 4668 1542}%
\special{pa 4668 1536}%
\special{pa 4670 1506}%
\special{pa 4676 1478}%
\special{fp}%
\special{pa 4694 1414}%
\special{pa 4708 1386}%
\special{pa 4724 1358}%
\special{pa 4726 1358}%
\special{fp}%
\special{pa 4770 1308}%
\special{pa 4790 1288}%
\special{pa 4816 1270}%
\special{pa 4820 1268}%
\special{fp}%
\special{pa 4880 1240}%
\special{pa 4904 1232}%
\special{pa 4934 1224}%
\special{pa 4942 1222}%
\special{fp}%
\special{pa 5006 1210}%
\special{pa 5032 1206}%
\special{pa 5064 1202}%
\special{pa 5070 1200}%
\special{fp}%
% DOT 0 0 3 0 Black Black
% 4 3000 1600 2900 1600 3100 1600 3100 1600
% 
\special{pn 4}%
\special{sh 1}%
\special{ar 3000 1600 16 16 0  6.28318530717959E+0000}%
\special{sh 1}%
\special{ar 2900 1600 16 16 0  6.28318530717959E+0000}%
\special{sh 1}%
\special{ar 3100 1600 16 16 0  6.28318530717959E+0000}%
\special{sh 1}%
\special{ar 3100 1600 16 16 0  6.28318530717959E+0000}%
% STR 2 0 3 0 Black Black
% 4 1400 1900 1400 2000 5 0 0 0
% $B_{1}$
\put(14.0000,-20.0000){\makebox(0,0){$B_{1}$}}%
% STR 2 0 3 0 Black Black
% 4 2400 1900 2400 2000 5 0 0 0
% $B_{2}$
\put(24.0000,-20.0000){\makebox(0,0){$B_{2}$}}%
% STR 2 0 3 0 Black Black
% 4 3700 1900 3700 2000 5 0 0 0
% $B_{i_{0}-1}$
\put(37.0000,-20.0000){\makebox(0,0){$B_{i_{0}-1}$}}%
% STR 2 0 3 0 Black Black
% 4 4600 1900 4600 2000 5 0 0 0
% $B_{i_{0}}$
\put(46.0000,-20.0000){\makebox(0,0){$B_{i_{0}}$}}%
% CIRCLE 2 0 0 0 Black Black
% 4 990 3000 990 3050 990 3050 990 3050
% 
\special{sh 1.000}%
\special{ia 990 3000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 990 3000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1090 3400 1090 3450 1090 3450 1090 3450
% 
\special{sh 1.000}%
\special{ia 1090 3400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1090 3400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1190 3000 1190 3050 1190 3050 1190 3050
% 
\special{sh 1.000}%
\special{ia 1190 3000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1190 3000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1390 3000 1390 3050 1390 3050 1390 3050
% 
\special{sh 1.000}%
\special{ia 1390 3000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1390 3000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1590 3000 1590 3050 1590 3050 1590 3050
% 
\special{sh 1.000}%
\special{ia 1590 3000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1590 3000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1290 3400 1290 3450 1290 3450 1290 3450
% 
\special{sh 1.000}%
\special{ia 1290 3400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1290 3400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1490 3400 1490 3450 1490 3450 1490 3450
% 
\special{sh 1.000}%
\special{ia 1490 3400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1490 3400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1790 3000 1790 3050 1790 3050 1790 3050
% 
\special{sh 1.000}%
\special{ia 1790 3000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1790 3000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1690 3400 1690 3450 1690 3450 1690 3450
% 
\special{sh 1.000}%
\special{ia 1690 3400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1690 3400 50 50  0.0000000  6.2831853}%
% LINE 0 0 3 0 Black Black
% 2 990 3000 1090 3400
% 
\special{pn 20}%
\special{pa 990 3000}%
\special{pa 1090 3400}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1190 3000 1290 3400
% 
\special{pn 20}%
\special{pa 1190 3000}%
\special{pa 1290 3400}%
\special{fp}%
% LINE 2 1 3 0 Black Black
% 2 1390 3000 1490 3400
% 
\special{pn 8}%
\special{pa 1390 3000}%
\special{pa 1490 3400}%
\special{da 0.070}%
% LINE 0 0 3 0 Black Black
% 2 1590 3000 1690 3400
% 
\special{pn 20}%
\special{pa 1590 3000}%
\special{pa 1690 3400}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1790 3000 1690 3400
% 
\special{pn 20}%
\special{pa 1790 3000}%
\special{pa 1690 3400}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1590 3000 1490 3400
% 
\special{pn 20}%
\special{pa 1590 3000}%
\special{pa 1490 3400}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1390 3000 1290 3400
% 
\special{pn 20}%
\special{pa 1390 3000}%
\special{pa 1290 3400}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1190 3000 1090 3400
% 
\special{pn 20}%
\special{pa 1190 3000}%
\special{pa 1090 3400}%
\special{fp}%
% CIRCLE 2 0 0 0 Black Black
% 4 2190 3000 2190 3050 2190 3050 2190 3050
% 
\special{sh 1.000}%
\special{ia 2190 3000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 2190 3000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 2290 3400 2290 3450 2290 3450 2290 3450
% 
\special{sh 1.000}%
\special{ia 2290 3400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 2290 3400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 2390 3000 2390 3050 2390 3050 2390 3050
% 
\special{sh 1.000}%
\special{ia 2390 3000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 2390 3000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 2590 3000 2590 3050 2590 3050 2590 3050
% 
\special{sh 1.000}%
\special{ia 2590 3000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 2590 3000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 2490 3400 2490 3450 2490 3450 2490 3450
% 
\special{sh 1.000}%
\special{ia 2490 3400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 2490 3400 50 50  0.0000000  6.2831853}%
% LINE 0 0 3 0 Black Black
% 2 2190 3000 2290 3400
% 
\special{pn 20}%
\special{pa 2190 3000}%
\special{pa 2290 3400}%
\special{fp}%
% LINE 2 1 3 0 Black Black
% 2 2390 3000 2490 3400
% 
\special{pn 8}%
\special{pa 2390 3000}%
\special{pa 2490 3400}%
\special{da 0.070}%
% LINE 0 0 3 0 Black Black
% 2 2590 3000 2490 3400
% 
\special{pn 20}%
\special{pa 2590 3000}%
\special{pa 2490 3400}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 2390 3000 2290 3400
% 
\special{pn 20}%
\special{pa 2390 3000}%
\special{pa 2290 3400}%
\special{fp}%
% CIRCLE 2 0 0 0 Black Black
% 4 3390 3000 3390 3050 3390 3050 3390 3050
% 
\special{sh 1.000}%
\special{ia 3390 3000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3390 3000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3490 3400 3490 3450 3490 3450 3490 3450
% 
\special{sh 1.000}%
\special{ia 3490 3400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3490 3400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3590 3000 3590 3050 3590 3050 3590 3050
% 
\special{sh 1.000}%
\special{ia 3590 3000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3590 3000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3790 3000 3790 3050 3790 3050 3790 3050
% 
\special{sh 1.000}%
\special{ia 3790 3000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3790 3000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3990 3000 3990 3050 3990 3050 3990 3050
% 
\special{sh 1.000}%
\special{ia 3990 3000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3990 3000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3690 3400 3690 3450 3690 3450 3690 3450
% 
\special{sh 1.000}%
\special{ia 3690 3400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3690 3400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3890 3400 3890 3450 3890 3450 3890 3450
% 
\special{sh 1.000}%
\special{ia 3890 3400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3890 3400 50 50  0.0000000  6.2831853}%
% LINE 0 0 3 0 Black Black
% 2 3390 3000 3490 3400
% 
\special{pn 20}%
\special{pa 3390 3000}%
\special{pa 3490 3400}%
\special{fp}%
% LINE 2 1 3 0 Black Black
% 2 3590 3000 3690 3400
% 
\special{pn 8}%
\special{pa 3590 3000}%
\special{pa 3690 3400}%
\special{da 0.070}%
% LINE 0 0 3 0 Black Black
% 2 3790 3000 3890 3400
% 
\special{pn 20}%
\special{pa 3790 3000}%
\special{pa 3890 3400}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 3990 3000 3890 3400
% 
\special{pn 20}%
\special{pa 3990 3000}%
\special{pa 3890 3400}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 3790 3000 3690 3400
% 
\special{pn 20}%
\special{pa 3790 3000}%
\special{pa 3690 3400}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 3590 3000 3490 3400
% 
\special{pn 20}%
\special{pa 3590 3000}%
\special{pa 3490 3400}%
\special{fp}%
% CIRCLE 2 0 0 0 Black Black
% 4 4390 3000 4390 3050 4390 3050 4390 3050
% 
\special{sh 1.000}%
\special{ia 4390 3000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 4390 3000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 4490 3400 4490 3450 4490 3450 4490 3450
% 
\special{sh 1.000}%
\special{ia 4490 3400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 4490 3400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 4590 3000 4590 3050 4590 3050 4590 3050
% 
\special{sh 1.000}%
\special{ia 4590 3000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 4590 3000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 4790 3000 4790 3050 4790 3050 4790 3050
% 
\special{sh 1.000}%
\special{ia 4790 3000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 4790 3000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 4690 3400 4690 3450 4690 3450 4690 3450
% 
\special{sh 1.000}%
\special{ia 4690 3400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 4690 3400 50 50  0.0000000  6.2831853}%
% LINE 2 1 3 0 Black Black
% 2 4390 3000 4490 3400
% 
\special{pn 8}%
\special{pa 4390 3000}%
\special{pa 4490 3400}%
\special{da 0.070}%
% LINE 2 1 3 0 Black Black
% 2 4590 3000 4690 3400
% 
\special{pn 8}%
\special{pa 4590 3000}%
\special{pa 4690 3400}%
\special{da 0.070}%
% LINE 2 1 3 0 Black Black
% 2 4790 3000 4690 3400
% 
\special{pn 8}%
\special{pa 4790 3000}%
\special{pa 4690 3400}%
\special{da 0.070}%
% LINE 2 1 3 0 Black Black
% 2 4590 3000 4490 3400
% 
\special{pn 8}%
\special{pa 4590 3000}%
\special{pa 4490 3400}%
\special{da 0.070}%
% SPLINE 0 0 3 0 Black Black
% 4 3195 2800 3395 2860 3595 3000 3595 3000
% 
\special{pn 20}%
\special{pa 3196 2800}%
\special{pa 3228 2808}%
\special{pa 3258 2816}%
\special{pa 3290 2824}%
\special{pa 3320 2832}%
\special{pa 3350 2842}%
\special{pa 3380 2854}%
\special{pa 3436 2882}%
\special{pa 3464 2900}%
\special{pa 3516 2936}%
\special{pa 3542 2956}%
\special{pa 3566 2976}%
\special{pa 3592 2998}%
\special{pa 3596 3000}%
\special{sp}%
% SPLINE 0 0 3 0 Black Black
% 6 3695 3400 3695 3000 3795 2880 4195 2880 4395 3000 4395 3000
% 
\special{pn 20}%
\special{pa 3696 3400}%
\special{pa 3688 3366}%
\special{pa 3682 3334}%
\special{pa 3676 3300}%
\special{pa 3672 3266}%
\special{pa 3664 3202}%
\special{pa 3662 3168}%
\special{pa 3664 3136}%
\special{pa 3666 3106}%
\special{pa 3672 3074}%
\special{pa 3678 3044}%
\special{pa 3690 3014}%
\special{pa 3702 2986}%
\special{pa 3720 2958}%
\special{pa 3738 2932}%
\special{pa 3762 2908}%
\special{pa 3786 2888}%
\special{pa 3812 2870}%
\special{pa 3840 2856}%
\special{pa 3870 2846}%
\special{pa 3902 2838}%
\special{pa 3934 2834}%
\special{pa 3968 2832}%
\special{pa 4002 2832}%
\special{pa 4036 2836}%
\special{pa 4070 2842}%
\special{pa 4102 2848}%
\special{pa 4136 2858}%
\special{pa 4200 2882}%
\special{pa 4230 2896}%
\special{pa 4286 2928}%
\special{pa 4312 2944}%
\special{pa 4340 2962}%
\special{pa 4392 2998}%
\special{pa 4396 3000}%
\special{sp}%
% SPLINE 2 1 3 0 Black Black
% 5 4695 3400 4695 3000 4795 2880 5065 2800 5065 2800
% 
\special{pn 8}%
\special{pn 8}%
\special{pa 4696 3400}%
\special{pa 4688 3366}%
\special{pa 4684 3338}%
\special{fp}%
\special{pa 4672 3272}%
\special{pa 4670 3266}%
\special{pa 4666 3234}%
\special{pa 4664 3208}%
\special{fp}%
\special{pa 4662 3142}%
\special{pa 4662 3136}%
\special{pa 4666 3106}%
\special{pa 4670 3078}%
\special{fp}%
\special{pa 4690 3014}%
\special{pa 4690 3014}%
\special{pa 4702 2986}%
\special{pa 4720 2958}%
\special{pa 4722 2958}%
\special{fp}%
\special{pa 4764 2908}%
\special{pa 4786 2888}%
\special{pa 4812 2870}%
\special{pa 4816 2868}%
\special{fp}%
\special{pa 4876 2840}%
\special{pa 4898 2832}%
\special{pa 4938 2822}%
\special{fp}%
\special{pa 5002 2810}%
\special{pa 5026 2806}%
\special{pa 5060 2802}%
\special{pa 5066 2800}%
\special{fp}%
% DOT 0 0 3 0 Black Black
% 4 2995 3200 2895 3200 3095 3200 3095 3200
% 
\special{pn 4}%
\special{sh 1}%
\special{ar 2996 3200 16 16 0  6.28318530717959E+0000}%
\special{sh 1}%
\special{ar 2896 3200 16 16 0  6.28318530717959E+0000}%
\special{sh 1}%
\special{ar 3096 3200 16 16 0  6.28318530717959E+0000}%
\special{sh 1}%
\special{ar 3096 3200 16 16 0  6.28318530717959E+0000}%
% STR 2 0 3 0 Black Black
% 4 1590 2600 1590 2700 5 0 0 0
% $B'_{2}$
\put(15.9000,-27.0000){\makebox(0,0){$B'_{2}$}}%
% VECTOR 0 0 3 0 Black Black
% 2 3000 2200 3000 2600
% 
\special{pn 20}%
\special{pa 3000 2200}%
\special{pa 3000 2600}%
\special{fp}%
\special{sh 1}%
\special{pa 3000 2600}%
\special{pa 3020 2534}%
\special{pa 3000 2548}%
\special{pa 2980 2534}%
\special{pa 3000 2600}%
\special{fp}%
% STR 2 0 3 0 Black Black
% 4 795 3100 795 3200 5 0 0 0
% $B'_{1}$
\put(7.9500,-32.0000){\makebox(0,0){$B'_{1}$}}%
% STR 2 0 3 0 Black Black
% 4 2590 2600 2590 2700 5 0 0 0
% $B'_{3}$
\put(25.9000,-27.0000){\makebox(0,0){$B'_{3}$}}%
% STR 2 0 3 0 Black Black
% 4 3390 2600 3390 2700 5 0 0 0
% $B'_{i_{0}-1}$
\put(33.9000,-27.0000){\makebox(0,0){$B'_{i_{0}-1}$}}%
% STR 2 0 3 0 Black Black
% 4 3990 2600 3990 2700 5 0 0 0
% $B'_{i_{0}}$
\put(39.9000,-27.0000){\makebox(0,0){$B'_{i_{0}}$}}%
% SPLINE 2 1 3 0 Black Black
% 5 1500 1800 1500 1400 1700 1200 2400 1400 2400 1400
% 
\special{pn 8}%
\special{pn 8}%
\special{pa 1500 1800}%
\special{pa 1494 1768}%
\special{pa 1490 1740}%
\special{fp}%
\special{pa 1480 1676}%
\special{pa 1478 1668}%
\special{pa 1474 1634}%
\special{pa 1472 1614}%
\special{fp}%
\special{pa 1470 1550}%
\special{pa 1470 1538}%
\special{pa 1472 1506}%
\special{pa 1476 1488}%
\special{fp}%
\special{pa 1492 1426}%
\special{pa 1494 1416}%
\special{pa 1508 1386}%
\special{pa 1518 1370}%
\special{fp}%
\special{pa 1554 1316}%
\special{pa 1562 1306}%
\special{pa 1598 1272}%
\special{fp}%
\special{pa 1646 1232}%
\special{pa 1664 1220}%
\special{pa 1690 1206}%
\special{pa 1702 1202}%
\special{fp}%
\special{pa 1762 1180}%
\special{pa 1776 1176}%
\special{pa 1806 1170}%
\special{pa 1822 1170}%
\special{fp}%
\special{pa 1886 1172}%
\special{pa 1926 1176}%
\special{pa 1948 1180}%
\special{fp}%
\special{pa 2010 1198}%
\special{pa 2018 1200}%
\special{pa 2048 1210}%
\special{pa 2068 1220}%
\special{fp}%
\special{pa 2126 1246}%
\special{pa 2142 1252}%
\special{pa 2174 1268}%
\special{pa 2182 1272}%
\special{fp}%
\special{pa 2238 1302}%
\special{pa 2270 1322}%
\special{pa 2292 1334}%
\special{fp}%
\special{pa 2348 1368}%
\special{pa 2366 1380}%
\special{pa 2398 1398}%
\special{pa 2400 1400}%
\special{fp}%
% SPLINE 0 0 3 0 Black Black
% 5 1500 3400 1500 3000 1700 2800 2400 3000 2400 3000
% 
\special{pn 20}%
\special{pa 1500 3400}%
\special{pa 1494 3368}%
\special{pa 1482 3300}%
\special{pa 1478 3268}%
\special{pa 1474 3234}%
\special{pa 1470 3202}%
\special{pa 1470 3138}%
\special{pa 1472 3106}%
\special{pa 1478 3076}%
\special{pa 1484 3044}%
\special{pa 1494 3016}%
\special{pa 1508 2986}%
\special{pa 1524 2958}%
\special{pa 1542 2930}%
\special{pa 1562 2906}%
\special{pa 1610 2858}%
\special{pa 1636 2838}%
\special{pa 1664 2820}%
\special{pa 1690 2806}%
\special{pa 1720 2792}%
\special{pa 1748 2782}%
\special{pa 1776 2776}%
\special{pa 1806 2770}%
\special{pa 1836 2768}%
\special{pa 1866 2768}%
\special{pa 1926 2776}%
\special{pa 1956 2782}%
\special{pa 1986 2790}%
\special{pa 2018 2800}%
\special{pa 2048 2810}%
\special{pa 2112 2838}%
\special{pa 2142 2852}%
\special{pa 2174 2868}%
\special{pa 2206 2886}%
\special{pa 2238 2902}%
\special{pa 2270 2922}%
\special{pa 2302 2940}%
\special{pa 2366 2980}%
\special{pa 2398 2998}%
\special{pa 2400 3000}%
\special{sp}%
% SPLINE 2 1 3 0 Black Black
% 4 2500 1800 2500 1400 2800 1200 2800 1200
% 
\special{pn 8}%
\special{pn 8}%
\special{pa 2500 1800}%
\special{pa 2494 1764}%
\special{pa 2490 1742}%
\special{fp}%
\special{pa 2480 1684}%
\special{pa 2476 1660}%
\special{pa 2472 1624}%
\special{fp}%
\special{pa 2466 1564}%
\special{pa 2466 1558}%
\special{pa 2468 1526}%
\special{pa 2470 1506}%
\special{fp}%
\special{pa 2482 1448}%
\special{pa 2484 1440}%
\special{pa 2494 1412}%
\special{pa 2506 1394}%
\special{fp}%
\special{pa 2544 1348}%
\special{pa 2546 1344}%
\special{pa 2568 1324}%
\special{pa 2590 1310}%
\special{fp}%
\special{pa 2640 1278}%
\special{pa 2648 1272}%
\special{pa 2680 1256}%
\special{pa 2692 1250}%
\special{fp}%
\special{pa 2746 1224}%
\special{pa 2776 1212}%
\special{pa 2800 1200}%
\special{fp}%
% SPLINE 0 0 3 0 Black Black
% 4 2500 3400 2500 3000 2800 2800 2800 2800
% 
\special{pn 20}%
\special{pa 2500 3400}%
\special{pa 2494 3364}%
\special{pa 2486 3330}%
\special{pa 2480 3294}%
\special{pa 2476 3260}%
\special{pa 2472 3224}%
\special{pa 2468 3192}%
\special{pa 2466 3158}%
\special{pa 2468 3126}%
\special{pa 2470 3096}%
\special{pa 2476 3068}%
\special{pa 2484 3040}%
\special{pa 2494 3012}%
\special{pa 2508 2988}%
\special{pa 2526 2966}%
\special{pa 2546 2944}%
\special{pa 2568 2924}%
\special{pa 2620 2888}%
\special{pa 2648 2872}%
\special{pa 2680 2856}%
\special{pa 2710 2840}%
\special{pa 2742 2826}%
\special{pa 2776 2812}%
\special{pa 2800 2800}%
\special{sp}%
\end{picture}%

\caption{Construction of $B'_{i}$}
\label{thm2.1fig2}
\end{center}
\end{figure}

\begin{claim}%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\label{cl2.2}
\begin{enumerate}
\item[{\upshape(i)}]
For each $i$ with $1\leq i\leq i_{0}$, we have $|V(B'_{i})\cap T|\geq |V(B'_{i})\cap S|$.
\item[{\upshape(ii)}]
For each $i$ with $1\leq i\leq i_{0}-1$,
\begin{enumerate}
\item[{\upshape(a)}]
$|V(B'_{i})|$ is even or $|V(B'_{i})|\geq 3$, and
\item[{\upshape(b)}]
$V(B'_{i})\cap T_{2}\not=\emptyset $ if $B'_{i}\simeq P_{3}$.
\qed
\end{enumerate}
\end{enumerate}
\end{claim}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

Suppose that $i_{0}=p$.
Then
\begin{align}
\label{neweq2.9}
|V(B'_{p})|\geq 5
\end{align}
by (\ref{neweq2.5}) and (B5).
Let $F_{1}=(F_{0}-(\bigcup _{1\leq i\leq p}V(B_{i})))\cup (\bigcup _{1\leq i\leq p}B'_{i})$.
Then by Claim~\ref{cl2.2}, (\ref{neweq2.9}), (\ref{neweq2.1}), (\ref{neweq2.2}) and (B5), $F_{1}$ is an $S$-central path-factor of $G$, and $B'_{i}\not\in \C^{(1)}_{3}(F_{1})$ for each $i~(1\leq i\leq p)$.
Since $B_{1}\in \C^{(1)}_{3}(F_{0})$, we have $|\C^{(1)}_{3}(F_{1})|<|\C^{(1)}_{3}(F_{0})|$, which contradicts the minimality of $|\C^{(1)}_{3}(F)|$.
Thus $2\leq i_{0}\leq p-1$.
Then by the definition of $i_{0}$, $t_{i_{0}}=1$.
Hence $B''_{i_{0}}=B_{i_{0}}\cup B'_{i_{0}}$ is a path of $G$ with $|V(B''_{i_{0}})\cap T|\geq |V(B''_{i_{0}})\cap S|$ (see Figure~\ref{thm2.1fig3}).
Set $F_{2}=(F_{0}-(\bigcup _{1\leq i\leq i_{0}}V(B_{i})))\cup (\bigcup _{1\leq i\leq i_{0}-1}B'_{i})\cup B''_{i_{0}}$.
Then by Claim~\ref{cl2.2}, (\ref{neweq2.1}) and (\ref{neweq2.2}), $F_{2}$ is an $S$-central path-factor of $G$, and $B'_{i}\not\in \C^{(1)}_{3}(F_{1})$ for each $i~(1\leq i\leq i_{0}-1)$.
Furthermore,
\begin{align}
|V(B''_{i_{0}})|=|V(B_{i_{0}})|+|V(B'_{i_{0}})\cap V(B_{i_{0}-1})|.\label{cl2.1-12}
\end{align}
Since $|V(B'_{i_{0}})\cap V(B_{i_{0}-1})|\geq 2$ by (\ref{neweq2.5}), this implies $|V(B''_{i_{0}})|\geq 5$, and hence we also have $B''_{i_{0}}\not\in \C^{(1)}_{3}(F_{1})$.
Thus $|\C^{(1)}_{3}(F_{2})|=|\C^{(1)}_{3}(F_{0})|$.

\begin{figure}
\begin{center}
%WinTpicVersion4.26
\unitlength 0.1in
\begin{picture}( 44.8500, 22.8200)(  5.8500,-34.5000)
% CIRCLE 2 0 0 0 Black Black
% 4 995 1400 995 1450 995 1450 995 1450
% 
\special{sh 1.000}%
\special{ia 996 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 996 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1095 1800 1095 1850 1095 1850 1095 1850
% 
\special{sh 1.000}%
\special{ia 1096 1800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1096 1800 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1195 1400 1195 1450 1195 1450 1195 1450
% 
\special{sh 1.000}%
\special{ia 1196 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1196 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1395 1400 1395 1450 1395 1450 1395 1450
% 
\special{sh 1.000}%
\special{ia 1396 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1396 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1595 1400 1595 1450 1595 1450 1595 1450
% 
\special{sh 1.000}%
\special{ia 1596 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1596 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1295 1800 1295 1850 1295 1850 1295 1850
% 
\special{sh 1.000}%
\special{ia 1296 1800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1296 1800 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1495 1800 1495 1850 1495 1850 1495 1850
% 
\special{sh 1.000}%
\special{ia 1496 1800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1496 1800 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1795 1400 1795 1450 1795 1450 1795 1450
% 
\special{sh 1.000}%
\special{ia 1796 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1796 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1695 1800 1695 1850 1695 1850 1695 1850
% 
\special{sh 1.000}%
\special{ia 1696 1800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1696 1800 50 50  0.0000000  6.2831853}%
% LINE 0 0 3 0 Black Black
% 2 995 1400 1095 1800
% 
\special{pn 20}%
\special{pa 996 1400}%
\special{pa 1096 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1195 1400 1295 1800
% 
\special{pn 20}%
\special{pa 1196 1400}%
\special{pa 1296 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1395 1400 1495 1800
% 
\special{pn 20}%
\special{pa 1396 1400}%
\special{pa 1496 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1595 1400 1695 1800
% 
\special{pn 20}%
\special{pa 1596 1400}%
\special{pa 1696 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1795 1400 1695 1800
% 
\special{pn 20}%
\special{pa 1796 1400}%
\special{pa 1696 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1595 1400 1495 1800
% 
\special{pn 20}%
\special{pa 1596 1400}%
\special{pa 1496 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1395 1400 1295 1800
% 
\special{pn 20}%
\special{pa 1396 1400}%
\special{pa 1296 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1195 1400 1095 1800
% 
\special{pn 20}%
\special{pa 1196 1400}%
\special{pa 1096 1800}%
\special{fp}%
% CIRCLE 2 0 0 0 Black Black
% 4 2195 1400 2195 1450 2195 1450 2195 1450
% 
\special{sh 1.000}%
\special{ia 2196 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 2196 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 2295 1800 2295 1850 2295 1850 2295 1850
% 
\special{sh 1.000}%
\special{ia 2296 1800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 2296 1800 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 2395 1400 2395 1450 2395 1450 2395 1450
% 
\special{sh 1.000}%
\special{ia 2396 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 2396 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 2595 1400 2595 1450 2595 1450 2595 1450
% 
\special{sh 1.000}%
\special{ia 2596 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 2596 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 2495 1800 2495 1850 2495 1850 2495 1850
% 
\special{sh 1.000}%
\special{ia 2496 1800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 2496 1800 50 50  0.0000000  6.2831853}%
% LINE 0 0 3 0 Black Black
% 2 2195 1400 2295 1800
% 
\special{pn 20}%
\special{pa 2196 1400}%
\special{pa 2296 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 2395 1400 2495 1800
% 
\special{pn 20}%
\special{pa 2396 1400}%
\special{pa 2496 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 2595 1400 2495 1800
% 
\special{pn 20}%
\special{pa 2596 1400}%
\special{pa 2496 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 2395 1400 2295 1800
% 
\special{pn 20}%
\special{pa 2396 1400}%
\special{pa 2296 1800}%
\special{fp}%
% CIRCLE 2 0 0 0 Black Black
% 4 3395 1400 3395 1450 3395 1450 3395 1450
% 
\special{sh 1.000}%
\special{ia 3396 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3396 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3495 1800 3495 1850 3495 1850 3495 1850
% 
\special{sh 1.000}%
\special{ia 3496 1800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3496 1800 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3595 1400 3595 1450 3595 1450 3595 1450
% 
\special{sh 1.000}%
\special{ia 3596 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3596 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3795 1400 3795 1450 3795 1450 3795 1450
% 
\special{sh 1.000}%
\special{ia 3796 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3796 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3995 1400 3995 1450 3995 1450 3995 1450
% 
\special{sh 1.000}%
\special{ia 3996 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3996 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3695 1800 3695 1850 3695 1850 3695 1850
% 
\special{sh 1.000}%
\special{ia 3696 1800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3696 1800 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3895 1800 3895 1850 3895 1850 3895 1850
% 
\special{sh 1.000}%
\special{ia 3896 1800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3896 1800 50 50  0.0000000  6.2831853}%
% LINE 0 0 3 0 Black Black
% 2 3395 1400 3495 1800
% 
\special{pn 20}%
\special{pa 3396 1400}%
\special{pa 3496 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 3595 1400 3695 1800
% 
\special{pn 20}%
\special{pa 3596 1400}%
\special{pa 3696 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 3795 1400 3895 1800
% 
\special{pn 20}%
\special{pa 3796 1400}%
\special{pa 3896 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 3995 1400 3895 1800
% 
\special{pn 20}%
\special{pa 3996 1400}%
\special{pa 3896 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 3795 1400 3695 1800
% 
\special{pn 20}%
\special{pa 3796 1400}%
\special{pa 3696 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 3595 1400 3495 1800
% 
\special{pn 20}%
\special{pa 3596 1400}%
\special{pa 3496 1800}%
\special{fp}%
% CIRCLE 2 0 0 0 Black Black
% 4 4395 1400 4395 1450 4395 1450 4395 1450
% 
\special{sh 1.000}%
\special{ia 4396 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 4396 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 4495 1800 4495 1850 4495 1850 4495 1850
% 
\special{sh 1.000}%
\special{ia 4496 1800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 4496 1800 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 4595 1400 4595 1450 4595 1450 4595 1450
% 
\special{sh 1.000}%
\special{ia 4596 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 4596 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 4795 1400 4795 1450 4795 1450 4795 1450
% 
\special{sh 1.000}%
\special{ia 4796 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 4796 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 4695 1800 4695 1850 4695 1850 4695 1850
% 
\special{sh 1.000}%
\special{ia 4696 1800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 4696 1800 50 50  0.0000000  6.2831853}%
% LINE 0 0 3 0 Black Black
% 2 4395 1400 4495 1800
% 
\special{pn 20}%
\special{pa 4396 1400}%
\special{pa 4496 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 4595 1400 4695 1800
% 
\special{pn 20}%
\special{pa 4596 1400}%
\special{pa 4696 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 4795 1400 4695 1800
% 
\special{pn 20}%
\special{pa 4796 1400}%
\special{pa 4696 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 4595 1400 4495 1800
% 
\special{pn 20}%
\special{pa 4596 1400}%
\special{pa 4496 1800}%
\special{fp}%
% SPLINE 2 1 3 0 Black Black
% 4 3200 1200 3400 1260 3600 1400 3600 1400
% 
\special{pn 8}%
\special{pn 8}%
\special{pa 3200 1200}%
\special{pa 3262 1216}%
\special{fp}%
\special{pa 3326 1232}%
\special{pa 3326 1232}%
\special{pa 3356 1242}%
\special{pa 3386 1254}%
\special{pa 3386 1254}%
\special{fp}%
\special{pa 3446 1284}%
\special{pa 3468 1300}%
\special{pa 3496 1318}%
\special{pa 3498 1320}%
\special{fp}%
\special{pa 3550 1360}%
\special{pa 3572 1376}%
\special{pa 3596 1398}%
\special{pa 3600 1400}%
\special{fp}%
% SPLINE 2 1 3 0 Black Black
% 6 3700 1800 3700 1400 3800 1280 4200 1280 4400 1400 4400 1400
% 
\special{pn 8}%
\special{pn 8}%
\special{pa 3700 1800}%
\special{pa 3694 1766}%
\special{pa 3690 1738}%
\special{fp}%
\special{pa 3678 1674}%
\special{pa 3676 1666}%
\special{pa 3672 1634}%
\special{pa 3672 1612}%
\special{fp}%
\special{pa 3668 1548}%
\special{pa 3668 1536}%
\special{pa 3672 1506}%
\special{pa 3676 1484}%
\special{fp}%
\special{pa 3692 1422}%
\special{pa 3694 1414}%
\special{pa 3708 1386}%
\special{pa 3720 1366}%
\special{fp}%
\special{pa 3760 1316}%
\special{pa 3766 1308}%
\special{pa 3790 1288}%
\special{pa 3810 1276}%
\special{fp}%
\special{pa 3868 1250}%
\special{pa 3876 1246}%
\special{pa 3908 1238}%
\special{pa 3930 1236}%
\special{fp}%
\special{pa 3994 1232}%
\special{pa 4006 1232}%
\special{pa 4040 1236}%
\special{pa 4056 1240}%
\special{fp}%
\special{pa 4120 1252}%
\special{pa 4140 1258}%
\special{pa 4180 1274}%
\special{fp}%
\special{pa 4238 1300}%
\special{pa 4294 1330}%
\special{fp}%
\special{pa 4348 1366}%
\special{pa 4396 1398}%
\special{pa 4400 1400}%
\special{fp}%
% SPLINE 2 1 3 0 Black Black
% 5 4700 1800 4700 1400 4800 1280 5070 1200 5070 1200
% 
\special{pn 8}%
\special{pn 8}%
\special{pa 4700 1800}%
\special{pa 4694 1766}%
\special{pa 4688 1736}%
\special{fp}%
\special{pa 4678 1672}%
\special{pa 4676 1666}%
\special{pa 4672 1634}%
\special{pa 4670 1608}%
\special{fp}%
\special{pa 4668 1542}%
\special{pa 4668 1536}%
\special{pa 4670 1506}%
\special{pa 4676 1478}%
\special{fp}%
\special{pa 4694 1414}%
\special{pa 4708 1386}%
\special{pa 4724 1358}%
\special{pa 4726 1358}%
\special{fp}%
\special{pa 4770 1308}%
\special{pa 4790 1288}%
\special{pa 4816 1270}%
\special{pa 4820 1268}%
\special{fp}%
\special{pa 4880 1240}%
\special{pa 4904 1232}%
\special{pa 4934 1224}%
\special{pa 4942 1222}%
\special{fp}%
\special{pa 5006 1210}%
\special{pa 5032 1206}%
\special{pa 5064 1202}%
\special{pa 5070 1200}%
\special{fp}%
% DOT 0 0 3 0 Black Black
% 4 3000 1600 2900 1600 3100 1600 3100 1600
% 
\special{pn 4}%
\special{sh 1}%
\special{ar 3000 1600 16 16 0  6.28318530717959E+0000}%
\special{sh 1}%
\special{ar 2900 1600 16 16 0  6.28318530717959E+0000}%
\special{sh 1}%
\special{ar 3100 1600 16 16 0  6.28318530717959E+0000}%
\special{sh 1}%
\special{ar 3100 1600 16 16 0  6.28318530717959E+0000}%
% STR 2 0 3 0 Black Black
% 4 1400 1900 1400 2000 5 0 0 0
% $B_{1}$
\put(14.0000,-20.0000){\makebox(0,0){$B_{1}$}}%
% STR 2 0 3 0 Black Black
% 4 2400 1900 2400 2000 5 0 0 0
% $B_{2}$
\put(24.0000,-20.0000){\makebox(0,0){$B_{2}$}}%
% STR 2 0 3 0 Black Black
% 4 3700 1900 3700 2000 5 0 0 0
% $B_{i_{0}-1}$
\put(37.0000,-20.0000){\makebox(0,0){$B_{i_{0}-1}$}}%
% STR 2 0 3 0 Black Black
% 4 4600 1900 4600 2000 5 0 0 0
% $B_{i_{0}}$
\put(46.0000,-20.0000){\makebox(0,0){$B_{i_{0}}$}}%
% CIRCLE 2 0 0 0 Black Black
% 4 990 3000 990 3050 990 3050 990 3050
% 
\special{sh 1.000}%
\special{ia 990 3000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 990 3000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1090 3400 1090 3450 1090 3450 1090 3450
% 
\special{sh 1.000}%
\special{ia 1090 3400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1090 3400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1190 3000 1190 3050 1190 3050 1190 3050
% 
\special{sh 1.000}%
\special{ia 1190 3000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1190 3000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1390 3000 1390 3050 1390 3050 1390 3050
% 
\special{sh 1.000}%
\special{ia 1390 3000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1390 3000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1590 3000 1590 3050 1590 3050 1590 3050
% 
\special{sh 1.000}%
\special{ia 1590 3000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1590 3000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1290 3400 1290 3450 1290 3450 1290 3450
% 
\special{sh 1.000}%
\special{ia 1290 3400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1290 3400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1490 3400 1490 3450 1490 3450 1490 3450
% 
\special{sh 1.000}%
\special{ia 1490 3400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1490 3400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1790 3000 1790 3050 1790 3050 1790 3050
% 
\special{sh 1.000}%
\special{ia 1790 3000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1790 3000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1690 3400 1690 3450 1690 3450 1690 3450
% 
\special{sh 1.000}%
\special{ia 1690 3400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1690 3400 50 50  0.0000000  6.2831853}%
% LINE 0 0 3 0 Black Black
% 2 990 3000 1090 3400
% 
\special{pn 20}%
\special{pa 990 3000}%
\special{pa 1090 3400}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1190 3000 1290 3400
% 
\special{pn 20}%
\special{pa 1190 3000}%
\special{pa 1290 3400}%
\special{fp}%
% LINE 2 1 3 0 Black Black
% 2 1390 3000 1490 3400
% 
\special{pn 8}%
\special{pa 1390 3000}%
\special{pa 1490 3400}%
\special{da 0.070}%
% LINE 0 0 3 0 Black Black
% 2 1590 3000 1690 3400
% 
\special{pn 20}%
\special{pa 1590 3000}%
\special{pa 1690 3400}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1790 3000 1690 3400
% 
\special{pn 20}%
\special{pa 1790 3000}%
\special{pa 1690 3400}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1590 3000 1490 3400
% 
\special{pn 20}%
\special{pa 1590 3000}%
\special{pa 1490 3400}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1390 3000 1290 3400
% 
\special{pn 20}%
\special{pa 1390 3000}%
\special{pa 1290 3400}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1190 3000 1090 3400
% 
\special{pn 20}%
\special{pa 1190 3000}%
\special{pa 1090 3400}%
\special{fp}%
% CIRCLE 2 0 0 0 Black Black
% 4 2190 3000 2190 3050 2190 3050 2190 3050
% 
\special{sh 1.000}%
\special{ia 2190 3000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 2190 3000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 2290 3400 2290 3450 2290 3450 2290 3450
% 
\special{sh 1.000}%
\special{ia 2290 3400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 2290 3400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 2390 3000 2390 3050 2390 3050 2390 3050
% 
\special{sh 1.000}%
\special{ia 2390 3000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 2390 3000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 2590 3000 2590 3050 2590 3050 2590 3050
% 
\special{sh 1.000}%
\special{ia 2590 3000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 2590 3000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 2490 3400 2490 3450 2490 3450 2490 3450
% 
\special{sh 1.000}%
\special{ia 2490 3400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 2490 3400 50 50  0.0000000  6.2831853}%
% LINE 0 0 3 0 Black Black
% 2 2190 3000 2290 3400
% 
\special{pn 20}%
\special{pa 2190 3000}%
\special{pa 2290 3400}%
\special{fp}%
% LINE 2 1 3 0 Black Black
% 2 2390 3000 2490 3400
% 
\special{pn 8}%
\special{pa 2390 3000}%
\special{pa 2490 3400}%
\special{da 0.070}%
% LINE 0 0 3 0 Black Black
% 2 2590 3000 2490 3400
% 
\special{pn 20}%
\special{pa 2590 3000}%
\special{pa 2490 3400}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 2390 3000 2290 3400
% 
\special{pn 20}%
\special{pa 2390 3000}%
\special{pa 2290 3400}%
\special{fp}%
% CIRCLE 2 0 0 0 Black Black
% 4 3390 3000 3390 3050 3390 3050 3390 3050
% 
\special{sh 1.000}%
\special{ia 3390 3000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3390 3000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3490 3400 3490 3450 3490 3450 3490 3450
% 
\special{sh 1.000}%
\special{ia 3490 3400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3490 3400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3590 3000 3590 3050 3590 3050 3590 3050
% 
\special{sh 1.000}%
\special{ia 3590 3000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3590 3000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3790 3000 3790 3050 3790 3050 3790 3050
% 
\special{sh 1.000}%
\special{ia 3790 3000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3790 3000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3990 3000 3990 3050 3990 3050 3990 3050
% 
\special{sh 1.000}%
\special{ia 3990 3000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3990 3000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3690 3400 3690 3450 3690 3450 3690 3450
% 
\special{sh 1.000}%
\special{ia 3690 3400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3690 3400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3890 3400 3890 3450 3890 3450 3890 3450
% 
\special{sh 1.000}%
\special{ia 3890 3400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3890 3400 50 50  0.0000000  6.2831853}%
% LINE 0 0 3 0 Black Black
% 2 3390 3000 3490 3400
% 
\special{pn 20}%
\special{pa 3390 3000}%
\special{pa 3490 3400}%
\special{fp}%
% LINE 2 1 3 0 Black Black
% 2 3590 3000 3690 3400
% 
\special{pn 8}%
\special{pa 3590 3000}%
\special{pa 3690 3400}%
\special{da 0.070}%
% LINE 0 0 3 0 Black Black
% 2 3790 3000 3890 3400
% 
\special{pn 20}%
\special{pa 3790 3000}%
\special{pa 3890 3400}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 3990 3000 3890 3400
% 
\special{pn 20}%
\special{pa 3990 3000}%
\special{pa 3890 3400}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 3790 3000 3690 3400
% 
\special{pn 20}%
\special{pa 3790 3000}%
\special{pa 3690 3400}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 3590 3000 3490 3400
% 
\special{pn 20}%
\special{pa 3590 3000}%
\special{pa 3490 3400}%
\special{fp}%
% CIRCLE 2 0 0 0 Black Black
% 4 4390 3000 4390 3050 4390 3050 4390 3050
% 
\special{sh 1.000}%
\special{ia 4390 3000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 4390 3000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 4490 3400 4490 3450 4490 3450 4490 3450
% 
\special{sh 1.000}%
\special{ia 4490 3400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 4490 3400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 4590 3000 4590 3050 4590 3050 4590 3050
% 
\special{sh 1.000}%
\special{ia 4590 3000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 4590 3000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 4790 3000 4790 3050 4790 3050 4790 3050
% 
\special{sh 1.000}%
\special{ia 4790 3000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 4790 3000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 4690 3400 4690 3450 4690 3450 4690 3450
% 
\special{sh 1.000}%
\special{ia 4690 3400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 4690 3400 50 50  0.0000000  6.2831853}%
% LINE 0 0 3 0 Black Black
% 2 4390 3000 4490 3400
% 
\special{pn 20}%
\special{pa 4390 3000}%
\special{pa 4490 3400}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 4590 3000 4690 3400
% 
\special{pn 20}%
\special{pa 4590 3000}%
\special{pa 4690 3400}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 4790 3000 4690 3400
% 
\special{pn 20}%
\special{pa 4790 3000}%
\special{pa 4690 3400}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 4590 3000 4490 3400
% 
\special{pn 20}%
\special{pa 4590 3000}%
\special{pa 4490 3400}%
\special{fp}%
% SPLINE 0 0 3 0 Black Black
% 4 3195 2800 3395 2860 3595 3000 3595 3000
% 
\special{pn 20}%
\special{pa 3196 2800}%
\special{pa 3228 2808}%
\special{pa 3258 2816}%
\special{pa 3290 2824}%
\special{pa 3320 2832}%
\special{pa 3350 2842}%
\special{pa 3380 2854}%
\special{pa 3436 2882}%
\special{pa 3464 2900}%
\special{pa 3516 2936}%
\special{pa 3542 2956}%
\special{pa 3566 2976}%
\special{pa 3592 2998}%
\special{pa 3596 3000}%
\special{sp}%
% SPLINE 0 0 3 0 Black Black
% 6 3695 3400 3695 3000 3795 2880 4195 2880 4395 3000 4395 3000
% 
\special{pn 20}%
\special{pa 3696 3400}%
\special{pa 3688 3366}%
\special{pa 3682 3334}%
\special{pa 3676 3300}%
\special{pa 3672 3266}%
\special{pa 3664 3202}%
\special{pa 3662 3168}%
\special{pa 3664 3136}%
\special{pa 3666 3106}%
\special{pa 3672 3074}%
\special{pa 3678 3044}%
\special{pa 3690 3014}%
\special{pa 3702 2986}%
\special{pa 3720 2958}%
\special{pa 3738 2932}%
\special{pa 3762 2908}%
\special{pa 3786 2888}%
\special{pa 3812 2870}%
\special{pa 3840 2856}%
\special{pa 3870 2846}%
\special{pa 3902 2838}%
\special{pa 3934 2834}%
\special{pa 3968 2832}%
\special{pa 4002 2832}%
\special{pa 4036 2836}%
\special{pa 4070 2842}%
\special{pa 4102 2848}%
\special{pa 4136 2858}%
\special{pa 4200 2882}%
\special{pa 4230 2896}%
\special{pa 4286 2928}%
\special{pa 4312 2944}%
\special{pa 4340 2962}%
\special{pa 4392 2998}%
\special{pa 4396 3000}%
\special{sp}%
% SPLINE 2 1 3 0 Black Black
% 5 4695 3400 4695 3000 4795 2880 5065 2800 5065 2800
% 
\special{pn 8}%
\special{pn 8}%
\special{pa 4696 3400}%
\special{pa 4688 3366}%
\special{pa 4684 3338}%
\special{fp}%
\special{pa 4672 3272}%
\special{pa 4670 3266}%
\special{pa 4666 3234}%
\special{pa 4664 3208}%
\special{fp}%
\special{pa 4662 3142}%
\special{pa 4662 3136}%
\special{pa 4666 3106}%
\special{pa 4670 3078}%
\special{fp}%
\special{pa 4690 3014}%
\special{pa 4690 3014}%
\special{pa 4702 2986}%
\special{pa 4720 2958}%
\special{pa 4722 2958}%
\special{fp}%
\special{pa 4764 2908}%
\special{pa 4786 2888}%
\special{pa 4812 2870}%
\special{pa 4816 2868}%
\special{fp}%
\special{pa 4876 2840}%
\special{pa 4898 2832}%
\special{pa 4938 2822}%
\special{fp}%
\special{pa 5002 2810}%
\special{pa 5026 2806}%
\special{pa 5060 2802}%
\special{pa 5066 2800}%
\special{fp}%
% DOT 0 0 3 0 Black Black
% 4 2995 3200 2895 3200 3095 3200 3095 3200
% 
\special{pn 4}%
\special{sh 1}%
\special{ar 2996 3200 16 16 0  6.28318530717959E+0000}%
\special{sh 1}%
\special{ar 2896 3200 16 16 0  6.28318530717959E+0000}%
\special{sh 1}%
\special{ar 3096 3200 16 16 0  6.28318530717959E+0000}%
\special{sh 1}%
\special{ar 3096 3200 16 16 0  6.28318530717959E+0000}%
% STR 2 0 3 0 Black Black
% 4 1590 2600 1590 2700 5 0 0 0
% $B'_{2}$
\put(15.9000,-27.0000){\makebox(0,0){$B'_{2}$}}%
% VECTOR 0 0 3 0 Black Black
% 2 3000 2200 3000 2600
% 
\special{pn 20}%
\special{pa 3000 2200}%
\special{pa 3000 2600}%
\special{fp}%
\special{sh 1}%
\special{pa 3000 2600}%
\special{pa 3020 2534}%
\special{pa 3000 2548}%
\special{pa 2980 2534}%
\special{pa 3000 2600}%
\special{fp}%
% STR 2 0 3 0 Black Black
% 4 795 3100 795 3200 5 0 0 0
% $B'_{1}$
\put(7.9500,-32.0000){\makebox(0,0){$B'_{1}$}}%
% STR 2 0 3 0 Black Black
% 4 2590 2600 2590 2700 5 0 0 0
% $B'_{3}$
\put(25.9000,-27.0000){\makebox(0,0){$B'_{3}$}}%
% STR 2 0 3 0 Black Black
% 4 3390 2600 3390 2700 5 0 0 0
% $B'_{i_{0}-1}$
\put(33.9000,-27.0000){\makebox(0,0){$B'_{i_{0}-1}$}}%
% STR 2 0 3 0 Black Black
% 4 3990 2600 3990 2700 5 0 0 0
% $B''_{i_{0}}$
\put(39.9000,-27.0000){\makebox(0,0){$B''_{i_{0}}$}}%
% SPLINE 2 1 3 0 Black Black
% 5 1500 1800 1500 1400 1700 1200 2400 1400 2400 1400
% 
\special{pn 8}%
\special{pn 8}%
\special{pa 1500 1800}%
\special{pa 1494 1768}%
\special{pa 1490 1740}%
\special{fp}%
\special{pa 1480 1676}%
\special{pa 1478 1668}%
\special{pa 1474 1634}%
\special{pa 1472 1614}%
\special{fp}%
\special{pa 1470 1550}%
\special{pa 1470 1538}%
\special{pa 1472 1506}%
\special{pa 1476 1488}%
\special{fp}%
\special{pa 1492 1426}%
\special{pa 1494 1416}%
\special{pa 1508 1386}%
\special{pa 1518 1370}%
\special{fp}%
\special{pa 1554 1316}%
\special{pa 1562 1306}%
\special{pa 1598 1272}%
\special{fp}%
\special{pa 1646 1232}%
\special{pa 1664 1220}%
\special{pa 1690 1206}%
\special{pa 1702 1202}%
\special{fp}%
\special{pa 1762 1180}%
\special{pa 1776 1176}%
\special{pa 1806 1170}%
\special{pa 1822 1170}%
\special{fp}%
\special{pa 1886 1172}%
\special{pa 1926 1176}%
\special{pa 1948 1180}%
\special{fp}%
\special{pa 2010 1198}%
\special{pa 2018 1200}%
\special{pa 2048 1210}%
\special{pa 2068 1220}%
\special{fp}%
\special{pa 2126 1246}%
\special{pa 2142 1252}%
\special{pa 2174 1268}%
\special{pa 2182 1272}%
\special{fp}%
\special{pa 2238 1302}%
\special{pa 2270 1322}%
\special{pa 2292 1334}%
\special{fp}%
\special{pa 2348 1368}%
\special{pa 2366 1380}%
\special{pa 2398 1398}%
\special{pa 2400 1400}%
\special{fp}%
% SPLINE 0 0 3 0 Black Black
% 5 1500 3400 1500 3000 1700 2800 2400 3000 2400 3000
% 
\special{pn 20}%
\special{pa 1500 3400}%
\special{pa 1494 3368}%
\special{pa 1482 3300}%
\special{pa 1478 3268}%
\special{pa 1474 3234}%
\special{pa 1470 3202}%
\special{pa 1470 3138}%
\special{pa 1472 3106}%
\special{pa 1478 3076}%
\special{pa 1484 3044}%
\special{pa 1494 3016}%
\special{pa 1508 2986}%
\special{pa 1524 2958}%
\special{pa 1542 2930}%
\special{pa 1562 2906}%
\special{pa 1610 2858}%
\special{pa 1636 2838}%
\special{pa 1664 2820}%
\special{pa 1690 2806}%
\special{pa 1720 2792}%
\special{pa 1748 2782}%
\special{pa 1776 2776}%
\special{pa 1806 2770}%
\special{pa 1836 2768}%
\special{pa 1866 2768}%
\special{pa 1926 2776}%
\special{pa 1956 2782}%
\special{pa 1986 2790}%
\special{pa 2018 2800}%
\special{pa 2048 2810}%
\special{pa 2112 2838}%
\special{pa 2142 2852}%
\special{pa 2174 2868}%
\special{pa 2206 2886}%
\special{pa 2238 2902}%
\special{pa 2270 2922}%
\special{pa 2302 2940}%
\special{pa 2366 2980}%
\special{pa 2398 2998}%
\special{pa 2400 3000}%
\special{sp}%
% SPLINE 2 1 3 0 Black Black
% 4 2500 1800 2500 1400 2800 1200 2800 1200
% 
\special{pn 8}%
\special{pn 8}%
\special{pa 2500 1800}%
\special{pa 2494 1764}%
\special{pa 2490 1742}%
\special{fp}%
\special{pa 2480 1684}%
\special{pa 2476 1660}%
\special{pa 2472 1624}%
\special{fp}%
\special{pa 2466 1564}%
\special{pa 2466 1558}%
\special{pa 2468 1526}%
\special{pa 2470 1506}%
\special{fp}%
\special{pa 2482 1448}%
\special{pa 2484 1440}%
\special{pa 2494 1412}%
\special{pa 2506 1394}%
\special{fp}%
\special{pa 2544 1348}%
\special{pa 2546 1344}%
\special{pa 2568 1324}%
\special{pa 2590 1310}%
\special{fp}%
\special{pa 2640 1278}%
\special{pa 2648 1272}%
\special{pa 2680 1256}%
\special{pa 2692 1250}%
\special{fp}%
\special{pa 2746 1224}%
\special{pa 2776 1212}%
\special{pa 2800 1200}%
\special{fp}%
% SPLINE 0 0 3 0 Black Black
% 4 2500 3400 2500 3000 2800 2800 2800 2800
% 
\special{pn 20}%
\special{pa 2500 3400}%
\special{pa 2494 3364}%
\special{pa 2486 3330}%
\special{pa 2480 3294}%
\special{pa 2476 3260}%
\special{pa 2472 3224}%
\special{pa 2468 3192}%
\special{pa 2466 3158}%
\special{pa 2468 3126}%
\special{pa 2470 3096}%
\special{pa 2476 3068}%
\special{pa 2484 3040}%
\special{pa 2494 3012}%
\special{pa 2508 2988}%
\special{pa 2526 2966}%
\special{pa 2546 2944}%
\special{pa 2568 2924}%
\special{pa 2620 2888}%
\special{pa 2648 2872}%
\special{pa 2680 2856}%
\special{pa 2710 2840}%
\special{pa 2742 2826}%
\special{pa 2776 2812}%
\special{pa 2800 2800}%
\special{sp}%
\end{picture}%

\caption{Construction of $B''_{i_{0}}$}
\label{thm2.1fig3}
\end{center}
\end{figure}

Set $k_{0}=\min\{k\mid B_{i_{0}}\in V(\P_{k})\}$, and write $B_{i_{0}}=A^{(k_{0})}_{j_{0}}$.
If $B_{p}\in V(\P_{k_{0}})$, then the fact that $B_{i_{0}}\not=B_{p}$ implies that $j_{0}\leq l_{k_{0}}-1$; if $B_{p}\not\in V(\P_{k_{0}})$, then the minimality of $k_{0}$ implies that $j_{0}\leq l_{k_{0}}-1$.
In either case, we have $j_{0}\leq l_{k_{0}}-1$.

\medskip
\noindent
\textbf{Case 1:} $j_{0}=1$.

\begin{sloppypar}
Since $B_{i_{0}}=A^{(k_{0})}_{1}$ and $i_{0}\geq 2$, $B_{1}\in \bigcup _{k_{0}+1\leq i\leq m}V(\P_{i})$.
In particular, $k_{0}\leq m-1$ and $\P_{k_{0}}$ is weakly admissible.
Hence $B_{i_{0}}\in \C^{(2)}_{5}(F_{0})\cup \C^{(1)}_{7}(F_{0})$.
This together with (\ref{cl2.1-12}) and (\ref{neweq2.5}) implies that $B''_{i_{0}}\in \C^{(2)}_{7}(F_{2})$ or $|V(B''_{i_{0}})|\geq 9$.
Thus the directed path $\P'_{k_{0}}=B''_{i_{0}}A^{(k_{0})}_{2}\cdots A^{(k_{0})}_{l_{k_{0}}}$ of $\D_{F_{2}}$ is strongly admissible.
Consequently $(\P_{1},\ldots,\P_{k_{0}-1},\P'_{k_{0}})$ is a complete path system with respect to $F_{2}$.
Since $k_{0}\leq m-1$ and $|V(\P_{k_{0}})|=|V(\P'_{k_{0}})|$, we see that $(|V(\P_{1})|,\ldots,|V(\P_{k_{0}-1})|,|V(\P'_{k_{0}})|)$ is lexicographically less than $(|V(\P_{1})|,\ldots,|V(\P_{k_{0}-1})|,|V(\P_{k_{0}})|,\ldots,|V(\P_{m})|)$, which contradicts the minimality of $(|V(\P_{1})|,\ldots,|V(\P_{m})|)$.
\end{sloppypar}

\medskip
\noindent
\textbf{Case 2:} $2\leq j_{0}\leq l_{k_{0}}-1$.

Since $B_{i_{0}}=A^{(k_{0})}_{j_{0}}$, $B_{i_{0}}\in \C^{(2)}_{3}(F_{0})\cup \C^{(1)}_{5}(F_{0})$.
This together with (\ref{cl2.1-12}) and (\ref{neweq2.5}) implies that $B''_{i_{0}}\in \C^{(2)}_{5}(F_{2})$ or $|V(B''_{i_{0}})|\geq 7$.
Thus the directed path $$\P'_{k_{0}}=B''_{i_{0}}A^{(k_{0})}_{j_{0}+1}A^{(k_{0})}_{j_{0}+2}\cdots A^{(k_{0})}_{l_{k_{0}}}$$ of $\D_{F_{2}}$ is admissible.
Consequently $(\P_{1},\ldots,\P_{k_{0}-1},\P'_{k_{0}})$ is a path system with respect to $F_{2}$.
By Claim~\ref{cl2.1}, the system can be extend to a complete path system $$(\P_{1},\ldots,\P_{k_{0}-1},\P'_{k_{0}},\Q_{1},\ldots,\Q_{\alpha })$$ with respect to $F_{2}$ (it is possible that $\alpha =0$).
Since $j_{0}\geq 2$, $|V(\P'_{k_{0}})|=l_{k_{0}}-j_{0}+1<l_{k_{0}}=|V(\P_{k_{0}})|$, and hence $(|V(\P_{1})|,\ldots,|V(\P_{k_{0}-1})|,|V(\P'_{k_{0}})|,|V(\Q_{1})|,\ldots,|V(\Q_{\alpha })|)$ is lexicographically less than $(|V(\P_{1})|,\ldots,|V(\P_{k_{0}-1})|,|V(\P_{k_{0}})|,\ldots,|V(\P_{m})|)$, which contradicts the minimality of $(|V(\P_{1})|,\ldots,|V(\P_{m})|)$.

This completes the proof of Theorem~\ref{thm2}.
\qed










%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Proof of Theorem~\ref{thm1}}\label{sec3}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

Let $G$ be as in Theorem~\ref{thm1}.
By assumption, we have $c_{1}(G)+\frac{2}{3}c_{3}(G)\leq \frac{4}{3}|\emptyset |+\frac{1}{3}=\frac{1}{3}$.
Hence $c_{1}(G)=c_{3}(G)=0$.

We now proceed by induction on $|V(G)|+|E(G)|$.
We may assume $V(G)\not=\emptyset $.
Note that if $E(G)=\emptyset $, then $c_{1}(G)=|V(G)|\geq 1$, which is a contradiction.
This means that the theorem holds for graphs $G$ with $E(G)=\emptyset $ in the sense that the assumption is not satisfied.
We henceforth assume that $E(G)\not=\emptyset $ and the theorem holds for graphs $G'$ with $|V(G')|+|E(G')|<|V(G)|+|E(G)|$.

Let $\S =\{X\subseteq V(G)\mid c_{1}(G-X)+c_{3}(G-X)\geq 1\}$.
Since $c_{1}(G-N_{G}(x))\geq 1$ for $x\in V(G)$, $\S \not=\emptyset $.
Set 
$$
\beta =\min_{X\in \S}\left\{\frac{4}{3}|X|+\frac{1}{3}-c_{1}(G-X)-\frac{2}{3}c_{3}(G-X)\right\}.
$$
%By the assumption of the theorem, $\beta \geq 0$.

\begin{claim}%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\label{cl3.0}
If $\beta \geq 2$, then $G$ has a $\{P_{2},P_{5}\}$-factor.
\end{claim}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\proof
Let $e\in E(G)$, and suppose that $\C_{1}(G-e)\cup \C_{3}(G-e)\not=\emptyset $.
Take $C\in \C_{1}(G-e)\cup \C_{3}(G-e)$.
Since $c_{1}(G)=c_{3}(G)=0$, $e$ joins a vertex in $V(C)$ and a vertex $y$ in $V(G)-V(C)$.
This implies $C\in \C_{1}(G-y)\cup \C_{3}(G-y)$, and hence $\frac{4}{3}|\{y\}|+\frac{1}{3}-(c_{1}(G-y)+\frac{2}{3}c_{3}(G-y))\leq \frac{4}{3}+\frac{1}{3}-\frac{2}{3}=1$, which contradicts the assumption that $\beta \geq 2$.
Thus $c_{1}(G-e)=c_{3}(G-e)=0$ for all $e\in E(G)$.
From the fact that $c_{1}(G-e)=\emptyset $ for all $e\in E(G)$, it follows that $d_{G}(x)\geq 2$ for all $x\in V(G)$.
Assume for the moment that $d_{G}(x)=2$ for all $x\in V(G)$.
Then each component of $G$ is a cycle.
Since $c_{3}(G)=0$, this implies that $G$ has a path-factor $F$ with $\C_{3}(F)=\emptyset $.
Hence by Fact~\ref{fact1}, $G$ has a $\{P_{2},P_{5}\}$-factor.
Thus we may assume that there exists $x_{0}\in V(G)$ such that $d_{G}(x_{0})\geq 3$.

Fix an edge $e^{*}=x_{0}y_{0}\in E(G)$ incident with $x_{0}$, and let $G'=G-e^{*}$.
By an assertion in the first paragraph of the proof the claim, $c_{1}(G')=c_{3}(G')=0$.
Let $X\subseteq V(G')$.
We show that $\frac{4}{3}|X|+\frac{1}{3}-c_{1}(G'-X)-\frac{2}{3}c_{3}(G'-X)\geq 0$.
We have $\frac{4}{3}|\emptyset |+\frac{1}{3}-c_{1}(G')-c_{3}(G')=\frac{1}{3}>0$.
Thus we may assume $X\not=\emptyset $.
Note that $c_{1}(G'-X)+c_{3}(G'-X)\leq c_{1}(G-X)+c_{3}(G-X)+2$, and hence
%$|(\C_{1}(G'-X)-\C_{1}(G-X)|+|(\C_{3}(G'-X)-\C_{3}(G-X)|\leq 2$, and hence
\begin{align}
c_{1}(G'-X)+\frac{2}{3}c_{3}(G'-X)\leq c_{1}(G-X)+\frac{2}{3}c_{3}(G-X)+2.\label{cl2.0-00}
\end{align}
Furthermore, if equality holds in (\ref{cl2.0-00}), then $x_{0},y_{0}\not\in X$ and $\{x_{0}\},\{y_{0}\}\in \C_{1}(G'-X)$.
If $c_{1}(G-X)+c_{3}(G-X)\geq 1$, then by the definition of $\beta $, $\frac{4}{3}|X|+\frac{1}{3}-c_{1}(G-X)-\frac{2}{3}c_{3}(G-X)\geq \beta \geq 2$ which, together with (\ref{cl2.0-00}), leads to $\frac{4}{3}|X|+\frac{1}{3}-(c_{1}(G'-X)+\frac{2}{3}c_{3}(G'-X))\geq \frac{4}{3}|X|+\frac{1}{3}-(c_{1}(G-X)+\frac{2}{3}c_{3}(G-X)+2)\geq \beta -2\geq 0$.
Thus we may assume that $c_{1}(G-X)+c_{3}(G-X)=0$.
By (\ref{cl2.0-00}), $c_{1}(G'-X)+c_{3}(G'-X)\leq c_{1}(G-X)+c_{3}(G-X)+2=2$.
By way of contradiction, suppose that $\frac{4}{3}|X|+\frac{1}{3}-(c_{1}(G'-X)+\frac{2}{3}c_{3}(G'-X))<0$.
Then $\frac{4}{3}|X|+\frac{1}{3}-2<0$.
Since $X\not=\emptyset $, this forces $|X|=1$ and $c_{1}(G'-X)+\frac{2}{3}c_{3}(G'-X)=2$.
Hence equality in (\ref{cl2.0-00}), which implies $\{x_{0}\}\in \C_{1}(G'-X)$.
Consequently $d_{G}(x_{0})\leq |X\cup \{y_{0}\}|=2$, which contradicts the fact that $d_{G}(x_{0})\geq 3$.
Thus we have $\frac{4}{3}|X|+\frac{1}{3}-c_{1}(G'-X)-\frac{2}{3}c_{3}(G'-X)\geq 0$ for all $X\subseteq V(G')$.
By the induction assumption, $G'$ has a $\{P_{2},P_{5}\}$-factor.
Therefore $G$ also has a $\{P_{2},P_{5}\}$-factor.
\qed

By Claim~\ref{cl3.0}, we may assume that $\beta \leq \frac{5}{3}$.

Let $S\in \S $ be a maximum set with $\frac{4}{3}|S|-c_{1}(G-S)-\frac{2}{3}c_{3}(G-S)+\frac{1}{3}=\beta $.
%Choose $S$ so that $|S|$ is as small as possible.

\begin{claim}%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\label{cl3.1}
Let $C$ be a component of $G-S$.
\begin{enumerate}
\item[{\upshape(i)}]
If $|V(C)|\not\in \{1,3\}$, then $C$ has a $\{P_{2},P_{5}\}$-factor.
\item[{\upshape(ii)}]
If $|V(C)|=3$, then $C$ is complete.
\end{enumerate}
\end{claim}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\proof
\begin{enumerate}
\item[{\upshape(i)}]
Suppose that $C$ has no $\{P_{2},P_{5}\}$-factor.
Then by the induction assumption, there exists a set $S'\subseteq V(C)$ with $\frac{4}{3}|S'|+\frac{1}{3}-c_{1}(C-S')-\frac{2}{3}c_{3}(C-S')<0$.
Set $S_{0}=S\cup S'$.
Since $\C_{1}(G-S_{0})=\C_{1}(G-S)\cup \C_{1}(C-S')$, $\C_{3}(G-S_{0})=\C_{3}(G-S)\cup \C_{3}(C-S')$ and $\C_{1}(C-S')\cup \C_{3}(C-S')\not=\emptyset $, we have $S_{0}\in \S $.
We also get $\frac{4}{3}|S_{0}|+\frac{1}{3}-c_{1}(G-S_{0})-\frac{2}{3}c_{3}(G-S_{0})=(\frac{4}{3}|S|+\frac{1}{3}-c_{1}(G-S)-\frac{2}{3}c_{3}(G-S))+(\frac{4}{3}|S'|-c_{1}(C-S')-\frac{2}{3}c_{3}(C-S'))<\beta  $.
This contradicts the definition of $\beta $.
\item[{\upshape(ii)}]
Suppose that $|V(C)|=3$ and $C$ is not complete (i.e., $C$ is a path of order three).
Let $x\in C$ be the vertex with $d_{C}(x)=2$.
Then $c_{1}(C-x)=2$ and $c_{3}(C-x)=0$.
Set $S_{1}=S\cup \{x\}$.
%Then $\frac{4}{3}|\{x\}|-c_{1}(C-x)-\frac{2}{3}c_{3}(C-x)=\frac{4}{3}-2=-\frac{2}{3}$.
Since $\C_{1}(G-S_{1})=\C_{1}(G-S)\cup \C_{1}(C-x)$, $\C_{3}(G-S_{1})=\C_{3}(G-S)-\{C\}$ and $\C_{1}(C-x)\not=\emptyset $, we have $S_{1}\in \S $.
We also get $\frac{4}{3}|S_{1}|+\frac{1}{3}-c_{1}(G-S_{1})-\frac{2}{3}c_{3}(G-S_{1})=(\frac{4}{3}|S|+\frac{4}{3})+\frac{1}{3}-(c_{1}(G-S)+2)-\frac{2}{3}(c_{3}(G-S)-1)=\beta $.
This contradicts the maximality of $S$.
\qed
\end{enumerate}

Set $T_{1}=\C_{1}(G-S)$, $T_{2}=\C_{3}(G-S)$ and $T=T_{1}\cup T_{2}$.
Now we construct a bipartite graph $H$ with bipartition $(S,T)$ by letting $uC\in E(H)~(u\in S,C\in T)$ if and only if $N_{G}(u)\cap V(C)\not=\emptyset $.

\begin{claim}%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\label{cl3.2}
The following hold.
\begin{enumerate}
\item[{\upshape(i)}]
$|T_{1}|+\frac{2}{3}|T_{2}|\leq \frac{4}{3}|S|+\frac{1}{3}$.
\item[{\upshape(ii)}]
$1\leq |S|\leq |T_{1}|+|T_{2}|$.
\item[{\upshape(iii)}]
For every $X\subseteq S_{0}$, either $|N_{H}(X)\cap T_{1}|+\frac{2}{3}|N_{H}(X)\cap T_{2}|\geq \frac{4}{3}|X|$ or $N_{H}(X)=T$.
\end{enumerate}
\end{claim}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\proof
\begin{enumerate}
\item[{\upshape(i)}]
By the assumption of the theorem, $|T_{1}|+\frac{2}{3}|T_{2}|=c_{1}(G-S)+\frac{2}{3}c_{3}(G-S)\leq \frac{4}{3}|S|+\frac{1}{3}$.
\item[{\upshape(ii)}]
Since $c_{1}(G)+c_{3}(G)=0$ and $c_{1}(G-S)+c_{3}(G-S)\geq 1$, $S\not=\emptyset $ (i.e., $|S|\geq 1$).
Since $\frac{4}{3}|S|+\frac{1}{3}-|T_{1}|-\frac{2}{3}|T_{2}|=\frac{4}{3}|S|+\frac{1}{3}-c_{1}(G-S)-\frac{2}{3}c_{3}(G-S)=\beta \leq \frac{5}{3}$, we get $|S|\leq \frac{3}{4}|T_{1}|+\frac{2}{4}|T_{2}|+1\leq \frac{3}{4}|T|+1<|T|+1$, and hence $|S|\leq |T|=|T_{1}|+|T_{2}|$.
\item[{\upshape(iii)}]
Suppose that there exists a set $X\subseteq S$ such that $|N_{H}(X)\cap T_{1}|+\frac{2}{3}|N_{H}(X)\cap T_{2}|<\frac{4}{3}|X|$ and $N_{H}(X)\not=T$.
Since $T-N_{H}(X)\subseteq \C_{1}(G-(S-X))\cup \C_{3}(G-(S-X))$ by the definition of $H$, we have $S-X\in \S$.
We also get $c_{1}(G-(S-X))+\frac{2}{3}c_{3}(G-(S-X))\geq (|T_{1}|-|N_{H}(X)\cap T_{1}|)+\frac{2}{3}(|T_{2}|-|N_{H}(X)\cap T_{2}|)=(c_{1}(G-S)+\frac{2}{3}c_{3}(G-S))-(|N_{H}(X)\cap T_{1}|+\frac{2}{3}|N_{H}(X)\cap T_{2}|)$.
Consequently $\frac{4}{3}|S-X|+\frac{1}{3}-c_{1}(G-(S-X))-\frac{2}{3}c_{3}(G-(S-X))\leq (\frac{4}{3}|S|+\frac{1}{3}-c_{1}(G-S)-\frac{2}{3}c_{3}(G-S))-(\frac{4}{3}|X|-|N_{H}(X)\cap T_{1}|-\frac{2}{3}|N_{H}(X)\cap T_{2}|)<\beta $, which contradicts the definition of $\beta $.
\qed
\end{enumerate}

By Claim~\ref{cl3.2} and Theorem~\ref{thm2}, $H$ has an $S$-central path-factor $F$ such that $V(A)\cap T_{2}\not=\emptyset $ for every $A\in \C_{3}(F)$.
For $A\in \C(F)$, let $U_{A}=V(A)\cap S$, $\L_{A,h}=V(A)\cap T_{h}~(h\in \{1,2\})$, and $\L_{A}=\L_{A,1}\cup \L_{A,2}$.
Let $G_{A}$ be the graph obtained from $G[U_{A}\cup (\bigcup _{C\in \L_{A}}V(C))]$ by deleting all edges of $G[U_{A}]$.

\begin{claim}%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\label{cl3.3}
For each $A\in \C(F)$, $G_{A}$ has a $\{P_{2},P_{5}\}$-factor.
\end{claim}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\proof
Since $A$ is a path of $H$, there exists a path $Q_{A}$ of $G_{A}$ such that $U_{A}\subseteq V(Q_{A})$ and $V(Q_{A})\cap V(C)\not=\emptyset $ for every $C\in \L_{A}$.
Choose $Q_{A}$ so that $|V(Q_{A})|$ is as large as possible.
Then for each $C\in \L_{A,2}$ (i.e., $C\in \L_{A}$ with $|V(C)|=3$), since $C$ is complete by Claim~\ref{cl3.1}(ii), it follows that
\begin{align*}
\mbox{either $V(C)\subseteq V(Q_{A})$ or $|V(C)\cap V(Q_{A})|=1$,}
\end{align*}
and
\begin{align*}
\mbox{if $C\in \L_{A}$ is an endvertex of the path $A$ of $H$, then $V(C)\subseteq V(Q_{A})$.}
\end{align*}
Recall that $\L_{A,2}\not=\emptyset $ if $|V(A)|=3$.
Consequently $|V(Q_{A})|\geq |V(A)|$ and, in the case where $|V(A)|\geq 3$, we have $|V(Q_{A})|=5$ or $7$.
Since $|V(A)|\geq 2$, this means that $|V(Q_{A})|\geq 2$ and $|V(Q_{A})|\not=3$.
Furthermore, for each $C\in \L_{A}$, $C-V(Q_{A})$ is either empty or a path of order two.
Therefore if we set $F_{A}=Q_{A}\cup (\bigcup _{C\in \L_{A}}(C-V(Q_{A})))$, then $F_{A}$ is a path-factor of $G_{A}$ with $\C_{3}(F_{A})=\emptyset $.
By Fact~\ref{fact1}, $G_{A}$ has a $\{P_{2},P_{5}\}$-factor.
\qed

By Claims~\ref{cl3.1}(i) and \ref{cl3.3}, $G$ has a $\{P_{2},P_{5}\}$-factor.

This completes the proof of Theorem~\ref{thm1}.











%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Examples}\label{sec4}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

In this section, we construct graphs having no $\{P_{2},P_{2k+1}\}$-factor.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\subsection{Graphs without $\{P_{2},P_{5}\}$-factor}\label{sec4.1}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

Let $n\geq 1$ be an integer.
Let $Q_{0}$ be a path of order $3$, and let $a$ be an endvertex of $Q_{0}$.
Let $Q_{1},\ldots,Q_{n}$ be disjoint paths of order $7$, and for each $i~(1\leq i\leq n)$, let $b_{i}$ be the center of $Q_{i}$.
Let $H_{n}$ denote the graph obtained from $\bigcup _{0\leq i\leq n}Q_{i}$ by joining $a$ to $b_{i}$ for every $i~(1\leq i\leq n)$ (see Figure~\ref{thm3.1fig1}).

\begin{figure}
\begin{center}
%WinTpicVersion4.26
\unitlength 0.1in
\begin{picture}( 28.5000, 15.8500)(  7.9000,-19.3500)
% CIRCLE 2 0 0 0 Black Black
% 4 995 1400 995 1450 995 1450 995 1450
% 
\special{sh 1.000}%
\special{ia 996 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 996 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1095 1800 1095 1850 1095 1850 1095 1850
% 
\special{sh 1.000}%
\special{ia 1096 1800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1096 1800 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1195 1400 1195 1450 1195 1450 1195 1450
% 
\special{sh 1.000}%
\special{ia 1196 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1196 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1395 1400 1395 1450 1395 1450 1395 1450
% 
\special{sh 1.000}%
\special{ia 1396 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1396 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1595 1400 1595 1450 1595 1450 1595 1450
% 
\special{sh 1.000}%
\special{ia 1596 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1596 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1295 1800 1295 1850 1295 1850 1295 1850
% 
\special{sh 1.000}%
\special{ia 1296 1800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1296 1800 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1495 1800 1495 1850 1495 1850 1495 1850
% 
\special{sh 1.000}%
\special{ia 1496 1800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1496 1800 50 50  0.0000000  6.2831853}%
% LINE 0 0 3 0 Black Black
% 2 995 1400 1095 1800
% 
\special{pn 20}%
\special{pa 996 1400}%
\special{pa 1096 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1195 1400 1295 1800
% 
\special{pn 20}%
\special{pa 1196 1400}%
\special{pa 1296 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1395 1400 1495 1800
% 
\special{pn 20}%
\special{pa 1396 1400}%
\special{pa 1496 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1595 1400 1495 1800
% 
\special{pn 20}%
\special{pa 1596 1400}%
\special{pa 1496 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1395 1400 1295 1800
% 
\special{pn 20}%
\special{pa 1396 1400}%
\special{pa 1296 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1195 1400 1095 1800
% 
\special{pn 20}%
\special{pa 1196 1400}%
\special{pa 1096 1800}%
\special{fp}%
% CIRCLE 2 0 0 0 Black Black
% 4 1790 1400 1790 1450 1790 1450 1790 1450
% 
\special{sh 1.000}%
\special{ia 1790 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1790 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1890 1800 1890 1850 1890 1850 1890 1850
% 
\special{sh 1.000}%
\special{ia 1890 1800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1890 1800 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1990 1400 1990 1450 1990 1450 1990 1450
% 
\special{sh 1.000}%
\special{ia 1990 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 1990 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 2190 1400 2190 1450 2190 1450 2190 1450
% 
\special{sh 1.000}%
\special{ia 2190 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 2190 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 2390 1400 2390 1450 2390 1450 2390 1450
% 
\special{sh 1.000}%
\special{ia 2390 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 2390 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 2090 1800 2090 1850 2090 1850 2090 1850
% 
\special{sh 1.000}%
\special{ia 2090 1800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 2090 1800 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 2290 1800 2290 1850 2290 1850 2290 1850
% 
\special{sh 1.000}%
\special{ia 2290 1800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 2290 1800 50 50  0.0000000  6.2831853}%
% LINE 0 0 3 0 Black Black
% 2 1790 1400 1890 1800
% 
\special{pn 20}%
\special{pa 1790 1400}%
\special{pa 1890 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1990 1400 2090 1800
% 
\special{pn 20}%
\special{pa 1990 1400}%
\special{pa 2090 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 2190 1400 2290 1800
% 
\special{pn 20}%
\special{pa 2190 1400}%
\special{pa 2290 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 2390 1400 2290 1800
% 
\special{pn 20}%
\special{pa 2390 1400}%
\special{pa 2290 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 2190 1400 2090 1800
% 
\special{pn 20}%
\special{pa 2190 1400}%
\special{pa 2090 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 1990 1400 1890 1800
% 
\special{pn 20}%
\special{pa 1990 1400}%
\special{pa 1890 1800}%
\special{fp}%
% CIRCLE 2 0 0 0 Black Black
% 4 2990 1400 2990 1450 2990 1450 2990 1450
% 
\special{sh 1.000}%
\special{ia 2990 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 2990 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3090 1800 3090 1850 3090 1850 3090 1850
% 
\special{sh 1.000}%
\special{ia 3090 1800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3090 1800 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3190 1400 3190 1450 3190 1450 3190 1450
% 
\special{sh 1.000}%
\special{ia 3190 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3190 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3390 1400 3390 1450 3390 1450 3390 1450
% 
\special{sh 1.000}%
\special{ia 3390 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3390 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3590 1400 3590 1450 3590 1450 3590 1450
% 
\special{sh 1.000}%
\special{ia 3590 1400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3590 1400 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3290 1800 3290 1850 3290 1850 3290 1850
% 
\special{sh 1.000}%
\special{ia 3290 1800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3290 1800 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3490 1800 3490 1850 3490 1850 3490 1850
% 
\special{sh 1.000}%
\special{ia 3490 1800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 3490 1800 50 50  0.0000000  6.2831853}%
% LINE 0 0 3 0 Black Black
% 2 2990 1400 3090 1800
% 
\special{pn 20}%
\special{pa 2990 1400}%
\special{pa 3090 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 3190 1400 3290 1800
% 
\special{pn 20}%
\special{pa 3190 1400}%
\special{pa 3290 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 3390 1400 3490 1800
% 
\special{pn 20}%
\special{pa 3390 1400}%
\special{pa 3490 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 3590 1400 3490 1800
% 
\special{pn 20}%
\special{pa 3590 1400}%
\special{pa 3490 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 3390 1400 3290 1800
% 
\special{pn 20}%
\special{pa 3390 1400}%
\special{pa 3290 1800}%
\special{fp}%
% LINE 0 0 3 0 Black Black
% 2 3190 1400 3090 1800
% 
\special{pn 20}%
\special{pa 3190 1400}%
\special{pa 3090 1800}%
\special{fp}%
% CIRCLE 2 0 0 0 Black Black
% 4 2300 800 2300 850 2300 850 2300 850
% 
\special{sh 1.000}%
\special{ia 2300 800 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 2300 800 50 50  0.0000000  6.2831853}%
% SPLINE 2 0 3 0 Black Black
% 5 2300 800 1500 1000 1300 1400 1300 1800 1300 1800
% 
\special{pn 8}%
\special{pa 2300 800}%
\special{pa 2266 802}%
\special{pa 2230 802}%
\special{pa 2194 804}%
\special{pa 2160 806}%
\special{pa 2124 806}%
\special{pa 2090 808}%
\special{pa 2054 810}%
\special{pa 2020 814}%
\special{pa 1986 816}%
\special{pa 1952 820}%
\special{pa 1920 824}%
\special{pa 1886 830}%
\special{pa 1822 842}%
\special{pa 1792 850}%
\special{pa 1760 858}%
\special{pa 1730 868}%
\special{pa 1702 878}%
\special{pa 1672 890}%
\special{pa 1644 902}%
\special{pa 1592 930}%
\special{pa 1566 948}%
\special{pa 1518 984}%
\special{pa 1474 1028}%
\special{pa 1454 1052}%
\special{pa 1434 1078}%
\special{pa 1398 1130}%
\special{pa 1382 1160}%
\special{pa 1368 1188}%
\special{pa 1354 1218}%
\special{pa 1342 1250}%
\special{pa 1330 1280}%
\special{pa 1320 1312}%
\special{pa 1312 1344}%
\special{pa 1294 1440}%
\special{pa 1292 1472}%
\special{pa 1288 1504}%
\special{pa 1288 1536}%
\special{pa 1286 1570}%
\special{pa 1288 1602}%
\special{pa 1288 1632}%
\special{pa 1294 1728}%
\special{pa 1298 1760}%
\special{pa 1300 1792}%
\special{pa 1300 1800}%
\special{sp}%
% SPLINE 2 0 3 0 Black Black
% 4 2300 800 2100 1200 2100 1800 2100 1800
% 
\special{pn 8}%
\special{pa 2300 800}%
\special{pa 2284 828}%
\special{pa 2248 884}%
\special{pa 2200 968}%
\special{pa 2184 998}%
\special{pa 2156 1054}%
\special{pa 2142 1084}%
\special{pa 2130 1112}%
\special{pa 2100 1202}%
\special{pa 2092 1232}%
\special{pa 2086 1262}%
\special{pa 2082 1292}%
\special{pa 2074 1356}%
\special{pa 2072 1386}%
\special{pa 2072 1418}%
\special{pa 2070 1450}%
\special{pa 2072 1482}%
\special{pa 2072 1514}%
\special{pa 2074 1548}%
\special{pa 2076 1580}%
\special{pa 2080 1612}%
\special{pa 2082 1646}%
\special{pa 2090 1710}%
\special{pa 2094 1744}%
\special{pa 2098 1776}%
\special{pa 2100 1800}%
\special{sp}%
% SPLINE 2 0 3 0 Black Black
% 5 2300 800 3100 1000 3300 1400 3300 1800 3300 1800
% 
\special{pn 8}%
\special{pa 2300 800}%
\special{pa 2336 802}%
\special{pa 2372 802}%
\special{pa 2406 804}%
\special{pa 2442 806}%
\special{pa 2476 806}%
\special{pa 2512 808}%
\special{pa 2546 810}%
\special{pa 2580 814}%
\special{pa 2614 816}%
\special{pa 2682 824}%
\special{pa 2778 842}%
\special{pa 2810 850}%
\special{pa 2840 858}%
\special{pa 2900 878}%
\special{pa 2956 902}%
\special{pa 2984 916}%
\special{pa 3010 930}%
\special{pa 3036 948}%
\special{pa 3084 984}%
\special{pa 3128 1028}%
\special{pa 3148 1052}%
\special{pa 3166 1078}%
\special{pa 3186 1104}%
\special{pa 3202 1130}%
\special{pa 3218 1160}%
\special{pa 3232 1188}%
\special{pa 3246 1218}%
\special{pa 3258 1250}%
\special{pa 3270 1280}%
\special{pa 3280 1312}%
\special{pa 3296 1376}%
\special{pa 3302 1408}%
\special{pa 3310 1472}%
\special{pa 3314 1536}%
\special{pa 3314 1602}%
\special{pa 3312 1632}%
\special{pa 3312 1664}%
\special{pa 3310 1696}%
\special{pa 3306 1728}%
\special{pa 3302 1792}%
\special{pa 3300 1800}%
\special{sp}%
% STR 2 0 3 0 Black Black
% 4 1000 1150 1000 1250 5 0 0 0
% $Q_{1}$
\put(10.0000,-12.5000){\makebox(0,0){$Q_{1}$}}%
% CIRCLE 2 0 0 0 Black Black
% 4 2300 600 2300 650 2300 650 2300 650
% 
\special{sh 1.000}%
\special{ia 2300 600 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 2300 600 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 2300 400 2300 450 2300 450 2300 450
% 
\special{sh 1.000}%
\special{ia 2300 400 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{pn 8}%
\special{ar 2300 400 50 50  0.0000000  6.2831853}%
% LINE 0 0 3 0 Black Black
% 2 2300 400 2300 800
% 
\special{pn 20}%
\special{pa 2300 400}%
\special{pa 2300 800}%
\special{fp}%
% STR 2 0 3 0 Black Black
% 4 1300 1900 1300 2000 5 0 0 0
% $b_{1}$
\put(13.0000,-20.0000){\makebox(0,0){$b_{1}$}}%
% STR 2 0 3 0 Black Black
% 4 2100 390 2100 490 5 0 0 0
% $Q_{0}$
\put(21.0000,-4.9000){\makebox(0,0){$Q_{0}$}}%
% STR 2 0 3 0 Black Black
% 4 2400 830 2400 930 5 0 0 0
% $a$
\put(24.0000,-9.3000){\makebox(0,0){$a$}}%
% DOT 0 0 3 0 Black Black
% 4 2800 1600 2700 1600 2600 1600 2600 1600
% 
\special{pn 4}%
\special{sh 1}%
\special{ar 2800 1600 16 16 0  6.28318530717959E+0000}%
\special{sh 1}%
\special{ar 2700 1600 16 16 0  6.28318530717959E+0000}%
\special{sh 1}%
\special{ar 2600 1600 16 16 0  6.28318530717959E+0000}%
\special{sh 1}%
\special{ar 2600 1600 16 16 0  6.28318530717959E+0000}%
% STR 2 0 3 0 Black Black
% 4 3300 1900 3300 2000 5 0 0 0
% $b_{n}$
\put(33.0000,-20.0000){\makebox(0,0){$b_{n}$}}%
% STR 2 0 3 0 Black Black
% 4 1800 1150 1800 1250 5 0 0 0
% $Q_{2}$
\put(18.0000,-12.5000){\makebox(0,0){$Q_{2}$}}%
% STR 2 0 3 0 Black Black
% 4 3000 1150 3000 1250 5 0 0 0
% $Q_{n}$
\put(30.0000,-12.5000){\makebox(0,0){$Q_{n}$}}%
% STR 2 0 3 0 Black Black
% 4 2100 1900 2100 2000 5 0 0 0
% $b_{2}$
\put(21.0000,-20.0000){\makebox(0,0){$b_{2}$}}%
\end{picture}%

\caption{Graph $H_{n}$}
\label{thm3.1fig1}
\end{center}
\end{figure}

Suppose that $H_{n}$ has a $\{P_{2},P_{5}\}$-factor $F$.
Since $Q_{0}$ does not have a $\{P_{2},P_{5}\}$-factor, $F$ contains $ab_{i}$ for some $i~(1\leq i\leq n)$.
Since $d_{F}(b_{i})\leq 2$, this requires that at least one of the components of $Q_{i}-b_{i}$ should have a $\{P_{2},P_{5}\}$-factor, which is impossible because each component of $Q_{i}-b_{i}$ is a path of order $3$.
Thus $H_{n}$ has no $\{P_{2},P_{5}\}$-factor.

\begin{lem}%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\label{lem4.1.1}
For all $X\subseteq V(H_{n})$, $c_{1}(H_{n}-X)+\frac{2}{3}c_{3}(H_{n}-X)\leq \frac{4}{3}|X|+\frac{2}{3}$.
\end{lem}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\proof
Let $X\subseteq V(H_{n})$.
Then we can verify that
\begin{align}
c_{1}(Q_{0}-X)+\frac{2}{3}c_{3}(Q_{0}-X)\leq \frac{4}{3}|V(Q_{0})\cap X|+\frac{2}{3}\label{sec4-1}
\end{align}
and
\begin{align}
c_{1}(Q_{i}-X)+\frac{2}{3}c_{3}(Q_{i}-X)\leq \frac{4}{3}|V(Q_{i})\cap X|\mbox{ for every }i~(1\leq i\leq n)\label{sec3-2}
\end{align}
Since every component $C$ of $H_{n}-X$ with $|V(C)|=1$ belongs to $\bigcup _{0\leq i\leq n}\C_{1}(Q_{i}-X)$, we have
\begin{align}
|\C_{1}(H_{n}-X)|=\sum _{0\leq i\leq n}|\C_{1}(Q_{i}-X)|-\left|\left(\bigcup _{0\leq i\leq n}\C_{1}(Q_{i}-X)\right)-\C_{1}(H_{n}-X)\right|.\label{sec4-3}
\end{align}
Furthermore,
\begin{align}
|\C_{3}(H_{n}-X)|\leq \sum _{0\leq i\leq n}|\C_{3}(Q_{i}-X)|+\left|\C_{3}(H_{n}-X)-\left(\bigcup _{0\leq i\leq n}\C_{3}(Q_{i}-X)\right)\right|.\label{sec4-4}
\end{align}

Let $C$ be a component of $H_{n}-X$ with $|V(C)|=3$ which does not belong to $\bigcup _{0\leq i\leq n}\C_{3}(Q_{i}-X)$.
Then $C$ intersects with at least two of the $Q_{i}~(0\leq i\leq n)$.
Since $|V(C)|=3$, $C$ contains a component of $Q_{i}-X$ of order $1$ for some $i~(0\leq i\leq n)$.
Since $C$ is arbitrary, this implies that
\begin{align}
\left|\C_{3}(H_{n}-X)-\left(\bigcup _{0\leq i\leq n}\C_{3}(Q_{i}-X)\right)\right|\leq \left|\left(\bigcup _{0\leq i\leq n}\C_{1}(Q_{i}-X)\right)-\C_{1}(H_{n}-X)\right|.\label{sec4-5}
\end{align}
By (\ref{sec4-1})--(\ref{sec4-5}),
\begin{align}
c_{1}(H_{n}-X)&+\frac{2}{3}c_{3}(H_{n}-X)\nonumber \\
&\leq \left(\sum _{0\leq i\leq n}|\C_{1}(Q_{i}-X)|-\left|\left(\bigcup _{0\leq i\leq n}\C_{1}(Q_{i}-X)\right)-\C_{1}(H_{n}-X)\right|\right)\nonumber \\
&\quad +\frac{2}{3}\left(\sum _{0\leq i\leq n}|\C_{3}(Q_{i}-X)|+\left|\C_{3}(H_{n}-X)-\left(\bigcup _{0\leq i\leq n}\C_{3}(Q_{i}-X)\right)\right|\right)\nonumber \\
&\leq \left(\sum _{0\leq i\leq n}|\C_{1}(Q_{i}-X)|-\left|\left(\bigcup _{0\leq i\leq n}\C_{1}(Q_{i}-X)\right)-\C_{1}(H_{n}-X)\right|\right)\nonumber \\
&\quad +\frac{2}{3}\left(\sum _{0\leq i\leq n}|\C_{3}(Q_{i}-X)|+\left|\left(\bigcup _{0\leq i\leq n}\C_{1}(Q_{i}-X)\right)-\C_{1}(H_{n}-X)\right|\right)\nonumber \\
&\leq \sum _{0\leq i\leq n}|\C_{1}(Q_{i}-X)|+\frac{2}{3}\sum _{0\leq i\leq n}|\C_{3}(Q_{i}-X)|\nonumber \\
&= \sum _{0\leq i\leq n}\left(c_{1}(Q_{i}-X)+\frac{2}{3}c_{3}(Q_{i}-X)\right)\nonumber \\
&\leq \frac{4}{3}\sum _{0\leq i\leq n}|V(Q_{i})\cap X|+\frac{2}{3}\nonumber \\
&= \frac{4}{3}|X|+\frac{2}{3}.\nonumber
\end{align}
Thus we get the desired conclusion.
\qed

From Lemma~\ref{lem4.1.1}, we get the following proposition, which implies that Theorem~\ref{thm1} is best possible.

\begin{prop}%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\label{prop3.1}
There exist infinitely many graphs $G$ having no $\{P_{2},P_{5}\}$-factor such that $c_{1}(G-X)+\frac{2}{3}c_{3}(G-X)\leq \frac{4}{3}|X|+\frac{2}{3}$ for all $X\subseteq V(G)$.
\end{prop}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\subsection{Graphs without $\{P_{2},P_{2k+1}\}$-factor for $k\geq 3$}\label{sec4.2}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

Let $k\geq 3$ be an integer with $k\equiv 0~(\mbox{mod }3)$, and write $k=3m$.
Let $n\geq 1$ be an integer.
Let $R_{0}$ be a complete graph of order $n$.
For each $i~(1\leq i\leq 2n+1)$, let $K_{i}$ be a complete graph of order $2m-1$, and let $R_{i}$ denote the graph obtained from $K_{i}$ by joining each vertex of the union of $2m+1$ disjoint paths of order $2$ to all vertices of $K_{i}$.
Let $H'_{n}=R_{0}+(\bigcup _{1\leq i\leq 2n+1}R_{i})$ (see Figure~\ref{thm3.2fig1}).

\begin{figure}
\begin{center}
%WinTpicVersion4.28b
{\unitlength 0.1in
\begin{picture}( 31.0000, 15.3500)( 12.5000,-21.8500)
% CIRCLE 2 0 0 0 Black Black
% 4 1400 2000 1400 2050 1400 2050 1400 2050
% 
\special{sh 1.000}%
\special{ia 1400 2000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 1400 2000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 1600 2000 1600 2050 1600 2050 1600 2050
% 
\special{sh 1.000}%
\special{ia 1600 2000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 1600 2000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 2200 2000 2200 2050 2200 2050 2200 2050
% 
\special{sh 1.000}%
\special{ia 2200 2000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 2200 2000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 2400 2000 2400 2050 2400 2050 2400 2050
% 
\special{sh 1.000}%
\special{ia 2400 2000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 2400 2000 50 50  0.0000000  6.2831853}%
% DOT 0 0 3 0 Black Black
% 4 2000 2000 1900 2000 1800 2000 1800 2000
% 
\special{pn 4}%
\special{sh 1}%
\special{ar 2000 2000 16 16 0  6.28318530717959E+0000}%
\special{sh 1}%
\special{ar 1900 2000 16 16 0  6.28318530717959E+0000}%
\special{sh 1}%
\special{ar 1800 2000 16 16 0  6.28318530717959E+0000}%
\special{sh 1}%
\special{ar 1800 2000 16 16 0  6.28318530717959E+0000}%
% LINE 1 0 3 0 Black Black
% 4 2200 2000 2400 2000 1600 2000 1400 2000
% 
\special{pn 13}%
\special{pa 2200 2000}%
\special{pa 2400 2000}%
\special{fp}%
\special{pa 1600 2000}%
\special{pa 1400 2000}%
\special{fp}%
% STR 2 0 3 0 Black Black
% 4 1900 1700 1900 1800 5 0 0 0
% $+$
\put(19.0000,-18.0000){\makebox(0,0){$+$}}%
% CIRCLE 2 0 0 0 Black Black
% 4 3200 2000 3200 2050 3200 2050 3200 2050
% 
\special{sh 1.000}%
\special{ia 3200 2000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 3200 2000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 3400 2000 3400 2050 3400 2050 3400 2050
% 
\special{sh 1.000}%
\special{ia 3400 2000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 3400 2000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 4000 2000 4000 2050 4000 2050 4000 2050
% 
\special{sh 1.000}%
\special{ia 4000 2000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 4000 2000 50 50  0.0000000  6.2831853}%
% CIRCLE 2 0 0 0 Black Black
% 4 4200 2000 4200 2050 4200 2050 4200 2050
% 
\special{sh 1.000}%
\special{ia 4200 2000 50 50  0.0000000  6.2831853}%
\special{pn 8}%
\special{ar 4200 2000 50 50  0.0000000  6.2831853}%
% DOT 0 0 3 0 Black Black
% 4 3800 2000 3700 2000 3600 2000 3600 2000
% 
\special{pn 4}%
\special{sh 1}%
\special{ar 3800 2000 16 16 0  6.28318530717959E+0000}%
\special{sh 1}%
\special{ar 3700 2000 16 16 0  6.28318530717959E+0000}%
\special{sh 1}%
\special{ar 3600 2000 16 16 0  6.28318530717959E+0000}%
\special{sh 1}%
\special{ar 3600 2000 16 16 0  6.28318530717959E+0000}%
% LINE 0 0 3 0 Black Black
% 4 4000 2000 4200 2000 3400 2000 3200 2000
% 
\special{pn 20}%
\special{pa 4000 2000}%
\special{pa 4200 2000}%
\special{fp}%
\special{pa 3400 2000}%
\special{pa 3200 2000}%
\special{fp}%
% STR 2 0 3 0 Black Black
% 4 3700 1700 3700 1800 5 0 0 0
% $+$
\put(37.0000,-18.0000){\makebox(0,0){$+$}}%
% DOT 0 0 3 0 Black Black
% 4 2900 1800 2800 1800 2700 1800 2700 1800
% 
\special{pn 4}%
\special{sh 1}%
\special{ar 2900 1800 16 16 0  6.28318530717959E+0000}%
\special{sh 1}%
\special{ar 2800 1800 16 16 0  6.28318530717959E+0000}%
\special{sh 1}%
\special{ar 2700 1800 16 16 0  6.28318530717959E+0000}%
\special{sh 1}%
\special{ar 2700 1800 16 16 0  6.28318530717959E+0000}%
% STR 2 0 3 0 Black Black
% 4 3700 2150 3700 2250 5 0 0 0
% $R_{2n+1}$
\put(37.0000,-22.5000){\makebox(0,0){$R_{2n+1}$}}%
% BOX 2 0 3 0 Black Black
% 2 1250 1450 2550 2150
% 
\special{pn 8}%
\special{pa 1250 1450}%
\special{pa 2550 1450}%
\special{pa 2550 2150}%
\special{pa 1250 2150}%
\special{pa 1250 1450}%
\special{pa 2550 1450}%
\special{fp}%
% BOX 2 0 3 0 Black Black
% 2 1300 1900 2500 2100
% 
\special{pn 8}%
\special{pa 1300 1900}%
\special{pa 2500 1900}%
\special{pa 2500 2100}%
\special{pa 1300 2100}%
\special{pa 1300 1900}%
\special{pa 2500 1900}%
\special{fp}%
% BOX 2 0 3 0 Black Black
% 2 3100 1900 4300 2100
% 
\special{pn 8}%
\special{pa 3100 1900}%
\special{pa 4300 1900}%
\special{pa 4300 2100}%
\special{pa 3100 2100}%
\special{pa 3100 1900}%
\special{pa 4300 1900}%
\special{fp}%
% BOX 2 0 3 0 Black Black
% 2 3050 1450 4350 2150
% 
\special{pn 8}%
\special{pa 3050 1450}%
\special{pa 4350 1450}%
\special{pa 4350 2150}%
\special{pa 3050 2150}%
\special{pa 3050 1450}%
\special{pa 4350 1450}%
\special{fp}%
% STR 2 0 3 0 Black Black
% 4 1900 2150 1900 2250 5 0 0 0
% $R_{1}$
\put(19.0000,-22.5000){\makebox(0,0){$R_{1}$}}%
% ELLIPSE 2 0 3 0 Black Black
% 4 2800 850 3200 1050 3200 1050 3200 1050
% 
\special{pn 8}%
\special{ar 2800 850 400 200  0.0000000  6.2831853}%
% STR 2 0 3 0 Black Black
% 4 2800 750 2800 850 5 0 0 0
% $R_{0}$
\put(28.0000,-8.5000){\makebox(0,0){$R_{0}$}}%
% ELLIPSE 2 0 3 0 Black Black
% 4 1900 1600 2400 1700 2400 1700 2400 1700
% 
\special{pn 8}%
\special{ar 1900 1600 500 100  0.0000000  6.2831853}%
% STR 2 0 3 0 Black Black
% 4 1900 1500 1900 1600 5 0 0 0
% $K_{1}$
\put(19.0000,-16.0000){\makebox(0,0){$K_{1}$}}%
% ELLIPSE 2 0 3 0 Black Black
% 4 3700 1600 4200 1700 4200 1700 4200 1700
% 
\special{pn 8}%
\special{ar 3700 1600 500 100  0.0000000  6.2831853}%
% STR 2 0 3 0 Black Black
% 4 3700 1500 3700 1600 5 0 0 0
% $K_{2n+1}$
\put(37.0000,-16.0000){\makebox(0,0){$K_{2n+1}$}}%
% LINE 2 0 3 0 Black Black
% 2 1250 1450 2430 770
% 
\special{pn 8}%
\special{pa 1250 1450}%
\special{pa 2430 770}%
\special{fp}%
% LINE 2 0 3 0 Black Black
% 2 3170 770 4350 1450
% 
\special{pn 8}%
\special{pa 3170 770}%
\special{pa 4350 1450}%
\special{fp}%
% LINE 2 0 3 0 Black Black
% 2 2550 1450 3160 940
% 
\special{pn 8}%
\special{pa 2550 1450}%
\special{pa 3160 940}%
\special{fp}%
% LINE 2 0 3 0 Black Black
% 2 2440 940 3050 1450
% 
\special{pn 8}%
\special{pa 2440 940}%
\special{pa 3050 1450}%
\special{fp}%
% STR 2 0 3 0 Black Black
% 4 2200 1100 2200 1200 5 0 0 0
% $+$
\put(22.0000,-12.0000){\makebox(0,0){$+$}}%
% STR 2 0 3 0 Black Black
% 4 3400 1100 3400 1200 5 0 0 0
% $+$
\put(34.0000,-12.0000){\makebox(0,0){$+$}}%
\end{picture}}%

\caption{Graph $H'_{n}$}
\label{thm3.2fig1}
\end{center}
\end{figure}

Since $|V(R_{i})|=2k+1$ and $R_{i}$ does not contain a path of order $2k+1$, $R_{i}$ has no $\{P_{2},P_{2k+1}\}$-factor.
Suppose that $H'_{n}$ has a $\{P_{2},P_{2k+1}\}$-factor $F$.
Then for each $i~(1\leq i\leq 2n+1)$, $F$ contains an edge joining $V(R_{i})$ and $V(R_{0})$.
Since $2n+1>2|V(R_{0})|$, this implies that there exists $x\in V(R_{0})$ such that $d_{F}(x)\geq 3$, which is a contradiction.
Thus $H'_{n}$ has no $\{P_{2},P_{2k+1}\}$-factor.

\begin{lem}%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\label{lem4.2.1}
For all $X\subseteq V(H'_{n})$, $\sum _{0\leq j\leq k-1}c_{2j+1}(H'_{n}-X)\leq \frac{4k+6}{8k+3}|X|+\frac{2k+3}{8k+3}$.
\end{lem}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\proof
Let $X\subseteq V(H'_{n})$.

\begin{claim}%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\label{claim4.2.1.1}
For each $i~(1\leq i\leq 2n+1)$, $\sum _{0\leq j\leq k-1}c_{2j+1}(R_{i}-X)\leq \frac{4k+6}{8k+3}|V(R_{i})\cap X|+\frac{2k+3}{8k+3}$.
\end{claim}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\proof
We first assume that $V(K_{i})\not\subseteq X$.
Then $R_{i}-X$ is connected.
Clearly we may assume that $\sum _{0\leq j\leq k-1}c_{2j+1}(R_{i}-X)=1$.
Then $|V(R_{i})\cap X|\geq 2$ because $|V(R_{i})|=2k+1$.
Hence $\sum _{0\leq j\leq k-1}c_{2j+1}(R_{i}-X)=1<\frac{4k+6}{8k+3}\cdot 2<\frac{4k+6}{8k+3}|V(R_{i})\cap X|+\frac{2k+3}{8k+3}$.
Thus we may assume that $V(K_{i})\subseteq X$.

Let $\alpha $ be the number of components of $R_{i}-V(K_{i})$ intersecting with $X$.
Since $\alpha \leq 2m+1$, we have $(8m+1)\alpha \leq (4m+2)(2m-1+\alpha )+2m+1$, and hence
$$
\alpha \leq \frac{4m+2}{8m+1}(2m-1+\alpha )+\frac{2m+1}{8m+1}=\frac{4k+6}{8k+3}(2m-1+\alpha )+\frac{2k+3}{8k+3}.
$$
Furthermore, $\sum _{0\leq j\leq k-1}c_{2j+1}(R_{i}-X)=c_{1}(R_{i}-X)\leq \alpha $ and $|V(R_{i})\cap X|=|V(K_{i})|+|(V(R_{i})-V(K_{i}))\cap X|\geq 2m-1+\alpha $.
Consequently we get $\sum _{0\leq j\leq k-1}c_{2j+1}(R_{i}-X)\leq \frac{4k+6}{8k+3}|V(R_{i})\cap X|+\frac{2k+3}{8k+3}$.
\qed

Assume for the moment that $V(R_{0})\not\subseteq X$.
Then $H'_{n}-X$ is connected.
Clearly we may assume that $\sum _{0\leq j\leq k-1}c_{2j+1}(H'_{n}-X)=1$.
Then $|X|\geq 2$ because $|V(H'_{n})|\geq 2k+1$.
Hence $\sum _{0\leq j\leq k-1}c_{2j+1}(H'_{n}-X)=1<\frac{4k+6}{8k+3}\cdot 2<\frac{4k+6}{8k+3}|X|+\frac{2k+3}{8k+3}$.
Thus we may assume that $V(R_{0})\subseteq X$.
Then clearly
\begin{align}
|&\C_{2j+1}(H'_{n}-X)|=\sum _{1\leq i\leq 2n+1}|\C_{2j+1}(R_{i}-X)|.\label{sec4-7}
\end{align}
By Claim~\ref{claim4.2.1.1} and (\ref{sec4-7}),
\begin{align*}
\sum _{0\leq j\leq k-1}c_{2j+1}(H'_{n}-X) &= \sum _{0\leq j\leq k-1}\left(\sum _{1\leq i\leq 2n+1}c_{2j+1}(R_{i}-X)\right)\\
&\leq \sum _{1\leq i\leq 2n+1}\left(\frac{4k+6}{8k+3}|V(R_{i})\cap X|+\frac{2k+3}{8k+3}\right)\\
&= \frac{4k+6}{8k+3}(|X|-|V(R_{0})|)+\frac{2k+3}{8k+3}(2n+1)\\
&= \frac{4k+6}{8k+3}(|X|-n)+\frac{2k+3}{8k+3}(2n+1)\\
&= \frac{4k+6}{8k+3}|X|+\frac{2k+3}{8k+3}.
\end{align*}
Thus we get the desired conclusion.
\qed

From Lemma~\ref{lem4.2.1}, we get the following proposition, which implies that if Conjecture~\ref{con1} is true, then the coefficient of $|X|$ in the conjecture is best possible.

\begin{prop}%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\label{prop3.2}
For an integer $k\geq 3$ with $k\equiv 0~(\mbox{mod }3)$, there exist infinitely many graphs $G$ having no $\{P_{2},P_{2k+1}\}$-factor such that $\sum _{0\leq i\leq k-1}c_{2i+1}(G-X)\leq \frac{4k+6}{8k+3}|X|+\frac{2k+3}{8k+3}$ for all $X\subseteq V(G)$.
\end{prop}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Concluding remarks}\label{sec5}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

In this paper, we prove that if a graph $G$ satisfies $c_{1}(G-X)+\frac{2}{3}c_{3}(G-X)\leq \frac{4}{3}|X|+\frac{1}{3}$ for all $X\subseteq V(G)$, then $G$ has a $\{P_{2},P_{5}\}$-factor.
The result naturally suggests the following problem:
For an integer $k\geq 3$, is there a number $\varepsilon _{k}>0$ such that if a graph $G$ satisfies $\sum _{0\leq j\leq k-1}c_{2j+1}(G-X)\leq \varepsilon _{k}|X|$ for all $X\subseteq V(G)$, then $G$ has a $\{P_{2},P_{2k+1}\}$-factor?
Recently, Egawa, Furuya and Ozeki~\cite{EFO} gave an affirmative solution to the problem with $\varepsilon _{k}=\frac{5}{6k^{2}}$.











\end{document}



